Issue 103

December 2019

WORKING PAPERS IN SCANDINAVIAN SYNTAX





Working Papers in Scandinavian Syntax is an electronic publication for current articles relating to the study of Scandinavian syntax. The articles appearing herein are previously unpublished reports of ongoing research activities and may subsequently appear, revised or unrevised, in other publications.

The WPSS homepage: http://project.sol.lu.se/grimm/working-papers-in-scandinavian-syntax/

The 104th volume of WPSS will be published in June 2020. Papers intended for publication should be submitted no later than May 1st, 2020.

Stockholm, December 2019,

Johan Brandtler, editor

Contact: Johan Brandtler Stockholm University Department of Swedish Language and Multilingualism 106 91 Stockholm, Sweden

E-mail: johan.brandtler@su.se

Working Papers in Scandinavian Syntax ISSN: 1100-097x Editor: Johan Brandtler, Stockholm University Editorial Board: Valéria Molnár, Lund University Halldór Ármann Sigurðsson, Lund University **WPSS 103**

Contents

A generalization on the Complementizer-Trace Effect		
from the intonational perspective		
Mayumi Hosono		
On the relative order of central sentence adverbs in	30	
the Insular Scandinavian Languages		
Ásgrímur Angantýsson		

A Generalization on the Complementizer-Trace Effect from the Intonational Perspective*

Mayumi Hosono

Keio University

Abstract

This paper presents a generalization on the *Complementizer-trace* effect from the intonational perspective on the basis of a comparative investigation collecting phonetic data from English and Finnish, in both of which the C-t context is acceptable to some speakers but unacceptable to others, as well as from Italian, Finland-Swedish and Dutch, in which the presence of an overt complementizer is either optional (Ita.), preferable (Fin.-Swe.) or obligatory (Dut.). The generalization on the C-t effect based on the data from English and Finnish is as follows: in the pitch gesture of the speaker who shows the C-t effect, the pitch is reset on the overt complementizer and the final pitch accent occurs within the complementizer clause. This generalization applies to an individual speaker, not to an individual language; the more native speakers of a language it applies to, that language is more likely to show the C-t effect. It is thus argued that the generalization here accounts for why the acceptability of the C-t context differs between languages as well as between the native speakers of a language. The C-t effect is accounted for in terms of the conflict of the pitch level on the element following the overt complementizer, i.e., the pitch should lower but actually rises on it. Based on the comparative study between Italian, Finland-Swedish and Dutch, all of which do not show the C-t effect, the optionality, preference and obligatoriness of an overt complementizer is discussed. It is argued that in the phonological/intonational environment where the pitch is more difficult to lower, the insertion of an overt complementizer is more preferable, and that the inserted complementizer acts as keeping the pitch level and enables the pitch to lower smoothly.

^{*} This is an ongoing research for which phonetic data collection and the pitch analysis are being done. Thanks to Johan Brandtler for his very helpful comments to improve this paper. Special thanks to Anders Holmberg for his many invaluable comments and suggestions to this work, and also acting as one of the informants. Thanks to William van der Wurff, Geoffrey Poole and Martha Young-Scholten, among others, for giving me helpful judgments data and participating in the recording. Thanks also to Johan Rooryck and Gunlög Joseffson for letting me know important facts relevant to the issue here on Dutch and Swedish respectively. I also would like to thank the informants who participated in the recording carried out in Lund University, Newcastle University and Leiden University. I take all responsibility in dealing with data and the way of interpreting them, as well as any other errors.

1. Introduction

The *Complementizer-trace* (C-t) effect illustrates a difference in acceptability between the extraction of a subject and that of other sentential elements from embedded clauses. The extraction, e.g. of a *wh*-object, from an embedded clause is acceptable, regardless of whether the complementizer *that* is present or not; see (1a-b). On the contrary, the extraction of a *wh*-subject from an embedded clause is not acceptable when the complementizer *that* overtly appears as illustrated in (2a), but it is acceptable when the complementizer does not appear as illustrated in (2b).

- a. What do you think [that John built __]?
 b. What do you think [Ø John built __]?
- (2) a. *Who do you think [that __ built the house]?
 b. Who do you think [\$\vee\$ __ built the house]?

In the tradition of generative syntax, the unacceptability of the extraction of a *wh*-subject from an embedded clause has long been attributed to a syntactic ill-formedness. The C-t effect has been accounted for in terms of movement of a *wh*-phrase across a complementizer. A *wh*-object can cross the overt complementizer *that* as in (1a), whereas a *wh*-subject cannot cross the overt complementizer as in (2a). In Chomsky (1981, 1986), the C-t effect was provided a representational account: *wh*-subject movement occurs in the same way as *wh*-object movement; the trace of a *wh*-subject is illicit due to the violation of the *Empty Category Principle* (ECP). Since Chomsky (1995), the C-t effect has been provided various derivational accounts. Chomsky (2015) attempts to provide an account for why the ECP can be violated when the complementizer is deleted. That is, C, a *phase* head (Chomsky 2008), has functional features such as φ -features, and they are all inherited by T in the course of a derivation (Richards 2007, Chomsky 2008). When C is deleted, the embedded CP phase boundary disappears, which enables a *wh*-subject to be involved in further syntactic operations.¹

It has been assumed since Chomsky (2001) that narrow syntax is uniform for all languages with surface difference confined to phonology. If the C-t effect arose from a syntactic ill-formedness, the C-t construction should be unacceptable for any speaker of any language, contrary to what has been reported in the literature.² Thus, experimental investigation on

¹ There has been so much literature on the C-t effect that I do not review them here. See, e.g. Bošković (2011) for an account of (the avoidance of) the C-t effect in terms of the affixation of a null complementizer to a matrix verb. See Pesetsky (2017) for a good summary of the theoretical accounts for the C-t effect in the history of Chomskyan generative syntax, and the references therein.

² See Kandybowicz (2006) for the references therein.

phonological/intonational factors will reveal detailed intonational properties of the constructions relevant to the C-t effect and provide an account for the ungrammaticality of the C-t effect. However, a long tradition exists in experimental phonetics that one should describe the intonation of a sentence as precisely as possible, where a 'sentence' means a 'grammatical sentence' (cf. Ladd 2008). If the ungrammaticality of an alleged syntactic phenomenon comes from phonological/intonational factors, it is necessary to make phonetic analysis of ungrammatical sentences and provide an account for their ungrammaticality based on phonetic data.

In this paper, I discuss the C-t effect based on an intonational data. This paper is organized as follows. Section 2 reviews Kandybowicz (2006), a previous phonological approach to the C-t effect. Section 3 introduces phonetic data of the constructions relevant to the C-t effect from English and Finnish, both of which show the C-t effect. In section 4, I propose a generalization on the C-t effect: in the pitch gesture of the speaker who shows the C-t effect, the pitch is reset on the overt complementizer and the final pitch accent occurs within the complementizer clause. I claim that the C-t effect arises from the conflict of the pitch level on the element following the overt complementizer, i.e., the pitch should lower but actually rises on it. Section 5 presents phonetic data of the constructions relevant to the C-t effect from the languages which do not show the C-t effect, i.e. Italian, Swedish and Dutch, and discusses the optionality, preference and obligatoriness of an overt complementizer. Section 6 concludes this paper.

2. Previous phonological approach to the C-t effect

Kandybowicz (2006) claims that the C-t effect is a phonological phenomenon, and that phonological properties are so different between languages that the C-t effect will not be given a unified account. His claim is based on data from English and Nupe, both of which show the C-t effect:

(3) a. *Who do you think [that __ built the house]?



The English complementizer is aligned at the left edge of the intermediate phrase composed of *that*-clausal elements, i.e. ...][_{intP} that ... (cf. Chomsky and Halle 1968). According to Kandybowicz, the C-t effect in English occurs when the following two conditions are satisfied: i) the overt complementizer and a trace are adjacent within the same prosodic phrase composed

of *that*-clausal elements; and ii) the overt complementizer is at the edge of the prosodic phrase.

On the contrary, the Nupe complementizer is aligned at the right edge of the intermediate phrase composed of main-clausal elements, i.e. ... ke][intP To determine that an overt complementizer is at the right edge of an intermediate phrase, Kandybowicz (2006) proposes the following criteria: i) an intonational break occurs after the complementizer; ii) the overt complementizer is lengthened; iii) the pitch is reset after the overt complementizer; and iv) phonological processes that should regularly occur are prevented. Then, the overt complementizer and a trace are not adjacent within the same prosodic phrase (composed of *that*-clausal elements), but the C-t effect still arises. On the assumption that the complement of the complementizer, TP, composes an intonation phrase, the C-t effect in Nupe is accounted for in terms of the violation of the condition that the edge of the intonation phrase, either [Spec,TP] or T, must be phonetically realized.

Kandybowicz's claims, however, can be criticized on both theoretical and empirical grounds. First, as long as one takes a phonological/intonational approach, one should not care about a trace. To be sure, the trace contributes to a semantic interpretation, e.g. in reconstruction, etc. But the trace does not have a phonetic form; it should not contribute to any prosodic phrasing to begin with. The approach is somewhat eclectic between syntactic theory and phonological theory; thus, a deeper conclusion cannot be drawn in his theory.

Secondly, contrary to his claim, it is not so important whether an overt complementizer is aligned at the left edge of the intermediate phrase composed of *that*-clausal elements as in English or at the right edge of the intermediate phrase composed of main-clausal elements as in Nupe. As we will see below, the prosodic phrasing of an overt complementizer differs between languages and even between the speakers of a language; despite such a great difference in the prosodic phrasing, the speakers who show the C-t effect have an intonational property consistently different from those who do not show the C-t effect.³

3. Intonational Properties of Languages that Show C-t Effect

From this section on, comparative experimental data is presented. Each of the test sentences was made in English and then systematically translated into the languages investigated. The test sentences comprised i) *wh*-object extraction with the complementizer, i.e. *what do you think that Bill wrote?*, ii) *wh*-object extraction without the complementizer, i.e. *what do you think Bill*

³ McFadden and Sundaresan (2018) attempts to provide an account for the C-t effect in terms of prosodic phrasing. Crucially, their prosodic phrasing of the English complementizer, i.e. ... that][intP ..., is wrong. See also Cowart (1997, 2003) for an experimental work on the C-t effect in terms of a native judgments survey, and Ritchart et al. (2016), who conclude based on their perception study that the prosodic approach to the C-t effect is not given any support.

wrote?, iii) *wh*-subject extraction with the complementizer, i.e. *who do you think that wrote the book*?, and iv) *wh*-subject extraction without the complementizer, i.e. *who do you think wrote the book*?.⁴ Some of the informants other than English native speakers corrected minor morphophonological differences such as Case, agreement, the expression of a *wh*-phrase, etc. Before the recording was done, the informants were asked to do native judgments of the test sentences, grading from OK, ?, to *. They were asked to read out all the test sentences, even if they felt some of them to be odd. They read out each sentence three times in appropriately rapid speech, in such a way as they speak in a real-life conversation. The recordings were made in a quiet small lecture room by the author herself. The voice of the informants was directly recorded into the author's laptop (LENOVO S21e), into which PRAAT speech processing software (Boersma and Weenink 1996) had been downloaded. The traces of fundamental frequency (F0) were computed by the autocorrelation method of PRAAT.

In the previous section, Kandybowicz' (2006) four criteria to determine the prosodic phrasing of an overt complementizer were presented. But as mentioned there, it depends on individual languages whether a complementizer is aligned at the left edge of the intermediate phrase composed of *that*-clausal elements, i.e. ...][intP C ..., or at the right edge of the intermediate phrase composed of main-clausal elements, i.e. ... C][intP As we will see in data of various languages, an overt complementizer is not necessarily lengthened; the pitch is not necessarily reset after an overt complementizer, either. We will also see that regular phonological processes do occur, illustrated by the contraction between an overt complementizer and the morpheme that either precedes or follows it. Thus in this paper, I simply decide the prosodic phrasing of an overt complementizer by the presence of an intonational break either before or after the complementizer, whether it is very long in some languages or quite short in others. When an intonational break is present before a complementizer, it is judged as prosodically phrased with its complement and located at the left edge of the intermediate phrase composed of *that*-clausal elements: ...][intP C ... When an intonational break is present after the complementizer, it is judged as prosodically phrased with a main clause and located at the right edge of the intermediate phrase composed of main-clausal elements: ... C][intP ... Below, the data on the extraction of a wh-object, in which a complementizer and a subject, i.e. a prominent argument, are adjacent to each other, is presented to confirm the prosodic phrasing of the complementizer, and then the data on the extraction of a *wh*-subject is presented.

⁴ The test sentences also comprised *wh*-subject extraction with the complementizer and an adverbial phrase following it, i.e. *who do you think that under no circumstances would betray you?*. This construction is claimed to mitigate the C-t effect (Bresnan 1977). The data on this construction was recorded to confirm the pitch gesture of native speakers and is not presented here.

First, the pitch patterns of an English native speaker who does not permit the C-t context (male, born in Essex, the UK) are presented:⁵



In the extraction of a *wh*-object, when the complementizer *that* is absent as in (4a), the entire sentence is pronounced as one intonational phrase. The pitch peak occurs on the sentence-initial *wh*-phrase, and the pitch continues to fall until the end of the entire sentence.⁶ When the complementizer is overt as in (4b), an intonational break occurs before the overt complementizer *that*. The overt complementizer is aligned at the left edge of the intermediate phrase composed of *that*-clausal elements, i.e. ...][intP that ... (Chomsky and Halle 1968, Kandybowicz 2006). The pitch is reset on the complementizer. The final pitch accent occurs within the *that*-clause, here on the subject *Bill*.

In the extraction of a *wh*-subject, when the complementizer is absent, the entire sentence is pronounced as one intonational phrase as in *wh*-object movement; see (5a). The pitch peak occurs on the sentence-initial *wh*-phrase, and the pitch continues to fall until the end of the

⁵ For the English intonational system, see Pierrehumbert (1980), Selkirk (1984, 1995), Bolinger (1998), Hirst (1998), Gussenhoven (2004), Ladd (2008) and Féry (2017), among others.

⁶ Cross-linguistically, the pitch falls at the end of a *wh*-question sentence (Bolinger 1978, Gordon 2016).

entire sentence. But when the complementizer is overt as illustrated in (5b), the pitch is reset on the complementizer and the final pitch accent occurs within the *that*-clause, here on the embedded verb *wrote*.



Secondly, the pitch patterns of an English native speaker who permits the C-t context (female, born in New Hampshire, the USA) are presented:





In the extraction of a *wh*-object, when the overt complementizer *that* is absent, the entire sentence is pronounced as one intonational phrase; see (6a). The pitch peak occurs on the sentence-initial *wh*-phrase, and the pitch continues to fall until the end of the entire sentence. But contrary to the speaker who does not permit the C-t context, when the complementizer is overt, the pitch is not reset on the complementizer; the pitch continues to lower, as illustrated in (6b).

In the extraction of a *wh*-subject, when the complementizer is absent, the entire sentence is pronounced as one intonational phrase; see (7a). The pitch peak occurs on the sentence-initial *wh*-phrase, and the pitch continues to fall until the end of the entire sentence. When the complementizer is overt, the pitch is not reset on the complementizer, contrary to the speaker who does not permit the C-t context; the pitch continues to lower; see (7b).





b. ^{OK}Who do you think that wrote the book?

The same tendency is observed in Finnish: some Finnish native speakers show the C-t effect, but others permit the C-t context. In the extraction of a *wh*-object, the complementizer *että* 'that' can be either absent (8a) or present (8b). But in the extraction of a *wh*-subject, the absence of the overt complementizer (9a) is more acceptable than its presence (9b) for some speakers, though others accept both patterns.⁷

- (8) a. Mitä luulet [Ø Billin kirjoittaneen]? [Fin.] what you-think Bill wrote 'What do you think Bill wrote?'
 - b. Mitä luulet [että Bill kirjoitti]?
 what you-think [that Bill wrote
 'What do you think that Bill wrote?'
- (9) a. Kenen luulet [Ø_kirjoittaneen kirjan]? [Fin.]
 who you-think wrote the book
 'Who do you think wrote the book?
 - b. ?Ketä sä luulet [että ____ kirjoitti kirjan]?
 who you think that wrote the book
 'Who do you think that wrote the book?'

First, the pitch patterns of a Finnish native speaker who does not permit the C-t context (male, born in Jämijärvi, Finland) are presented:⁸

⁷ The form *kirjoittaneen* is past participle, and the form *kirjoitti* is past tense.

⁸ See Iivonen (1998), Suomi et al. (2008) and Nakai et al. (2009) for the Finnish intonational system.



Mitä luulet Billin kirjoittaneen? ('what do you think Bill wrote?') (10) a.

b. Mitä luulet että Bill kirjoitti? ('what do you think that Bill wrote?')



In the extraction of a wh-object as illustrated in (10a), when the complementizer että 'that' is absent, the entire sentence is pronounced as one intonational phrase. The pitch peak occurs on the sentence-initial wh-phrase, and the pitch continues to fall until the end of the entire sentence. When the complementizer is overt as illustrated in (10b), an intonational break occurs before the overt complementizer että. In the same way as in English, the overt complementizer is aligned at the left edge of the intermediate phrase composed of että-clausal elements, i.e. ...][intP että The pitch is reset on the complementizer and the final pitch accent occurs within the että-clause, here on the subject Bill.

The pitch patterns of the construction of *wh*-subject extraction are presented below:



Kenen luulet kirjoittaneen kirjan? ('who do you think wrote the book?') (11) a.



b. ?Ketä sä luulet että kirjoitti kirjan? ('who do you think that wrote the book?')

When the complementizer is absent, the entire sentence is pronounced as one intonational phrase as in *wh*-object extraction; see (11a). The pitch peak occurs on the sentence-initial *wh*-phrase, and the pitch continues to fall until the end of the entire sentence. But when the complementizer is overt as in (11b), the pitch is reset on the complementizer and the final pitch accent occurs within the *että*-clause, here on the embedded object *kirjan* 'the book'.

Next, the pitch patterns of a Finnish native speaker who permits the C-t context (male, born in Turku, Finland) are presented:



b. Mitä luulet että Billin kirjoittaneen? ('what do you think that Bill wrote?')



In the extraction of a *wh*-object, when the complementizer *että* 'that' is absent, the entire sentence is pronounced as one intonational phrase; see (12a). The pitch peak occurs on the sentence-initial *wh*-phrase, and the pitch continues to fall until the end of the entire sentence. But contrary to the native Finnish speaker who does not permit the C-t context, when the complementizer is overt, the pitch is not reset on the complementizer; the pitch continues to lower, as illustrated in (12b). The first vowel *e*- of *että* is combined with the last consonant *-t* of the preceding verb *luulet* 'you-think', which produces a word-like unit, *luulet-että* [lu:letetta]. This contraction causes the complementizer *että* to be aligned at the right edge of the intermediate phrase composed of main-clausal elements, i.e. ... (luulet-)että][intP ... , with an intonational break present after the complementizer.

In the extraction of a *wh*-subject, when the complementizer is absent, the entire sentence is pronounced as one intonational phrase; see (13a). The pitch peak occurs on the sentence-initial *wh*-phrase, and the pitch continues to fall until the end of the entire sentence. When the complementizer is overt, the pitch is not reset on the complementizer, contrary to the speaker who does not permit the C-t context; the pitch continues to lower; see (13b). Here too, the first vowel *e*- of the complementizer *että* is combined with the last consonant *-t* of the preceding verb *luulet*, which produces a word-like unit, *luulet-että* [lu:letetta] and causes the complementizer *että* to be aligned at the right edge of the intermediate phrase composed of main-clausal elements, i.e. ... (luulet-)että][intP







4. Generalizing the C-t Effect from an Intonational Perspective

Based on English and Finnish, we have got a general description on the C-t effect. In the pitch gesture of the speaker who shows the C-t effect, the pitch is reset on the overt complementizer and the final pitch accent occurs within the complementizer clause, whereas in the pitch gesture of the speaker who permits the C-t context, the pitch is not reset on the overt complementizer and continues to lower. The generalization on the C-t effect is illustrated as follows:

(14) The C-t effect:

*...... Č t

The pitch is reset on the complementizer, which is indicated by \nearrow above C; the final pitch peak occurs within the complementizer clause, which is indicated by H*. The speaker who shows this pitch gesture does not permit the C-t context.

Note that generalization (14) applies to an individual speaker, not to an individual language. The pitch gesture and the pitch level differ depending on individual speakers. The more native speakers of a language (14) applies to, that language is more likely to show the C-t effect. When (14) applies to most of the speakers of a language, that language is said to show the C-t effect but can contain some exceptional speakers to whom (14) does not apply and who permit the C-t context. When (14) does not apply to most speakers of a language, that language is said not to show the C-t effect but can contain some exceptional speakers of a language, that language is said not to show the C-t effect but can contain some exceptional speakers to whom (14) applies and who does not permit the C-t context. The generalization here thus provides an account for why the acceptability of the C-t context differs between languages as well as between the native speakers of a language.

From the intonational perspective, the difference in the acceptability of the C-t context is accounted for as follows. In the pitch gesture of the speakers who do not permit the C-t context, the pitch is reset on the overt complementizer and the final pitch accent occurs within the complementizer clause. When the pitch is reset, the pitch rises on the overt complementizer and continues to rise towards the final pitch peak. In the extraction of a *wh*-object as illustrated in (4b) and (10b), a phonologically prominent element, i.e. a subject, is adjacent to the complementizer. The high pitch that occurs on the complementizer continues to rise and the subject. But in the extraction of a *wh*-subject, a verbal head is adjacent to the complementizer. The verbal head is not phonologically prominent in the

unmarked case, unless it is focused.⁹ The pitch should lower on the verbal head, but since it immediately follows the complementizer, the pitch actually rises on it. The conflict of the pitch level occurs on the verbal head, which causes the C-t effect; see (5b) and (11b). On the contrary, in the pitch gesture of the speakers who permit the C-t context, the pitch is not reset on the overt complementizer. The pitch continues to lower through the complementizer to the following sentential element, whether the latter is a nominal (6b,12b) or a verbal head (7b,13b). No conflict of the pitch level arises in the entire falling pitch, and the C-t effect does not occur. The C-t effect is thus accounted for in terms of the conflict of the pitch level on the element following the overt complementizer that should lower but actually rises on it:¹⁰

(15) Conflict of the pitch level:



Kandybowicz (2006) gives various constructions relevant to the C-t effect in English:

- (16) a. The author that the publisher predicts *that __ will be adored
 - b. It was John that the author told us *that __ had plagiarized her book.
 - c. I wrote more books than I estimated *that ____ would be written.
 - d. Who do you think ?th't wrote Barriers?
 - e. Who do you suppose ?that'll leave early?
 - f. Who do you think ^{OK} that after years and years of cheating death _____ finally died?
 - g. Who does John doubt whether and Bill suspect ^{OK}that ____ cheated on the exam?

The C-t effect occurs in embedded relative clauses (16a), cleft construction (16b) and comparative construction (16c). The pitch peak occurs on *the author* (16a), on *John* (16b) and on *more books* (16c) respectively, and then lowers. The verbal head (here the Aux) that is not phonologically prominent is adjacent to the (second) complementizer. In the same way as in the extraction of a *wh*-subject, the conflict of the pitch level occurs on the verbal head on which the

⁹ This is confirmed by much literature on phonology, e.g. Selkirk (1996), on syntax, e.g. Cinque (1993), and on information structure, e.g. Gundel (1988).

¹⁰ Anders Holmberg (p.c.) suggests that the cause and effect of the argument here might be the other way round. That is, the presence of the overt complementizer might cause an exceptional high pitch, thus the pitch resetting, on it in the C-t context. But it is not the case that the pitch is reset only in *wh*-subject extraction: the pitch is reset on the overt complementizer also in *wh*-object extraction as illustrated in (4b). It is thus plausible that the pitch resetting on the overt complementizer causes the conflict of the pitch level on the following element in the C-t context, as argued here.

pitch should lower but actually rises after the (second) overt complementizer.¹¹ It is reported that when the complementizer is reduced/unstressed (16d), and when the complementizer is contracted with the Aux crossing a subject trace (16e), the C-t effect is mitigated. It is expected that the pitch is lower on the reduced/contracted complementizer than on the full form of the complementizer; the pitch level will not conflict on the non-prominent verbal head following the complementizer. When an adverbial phrase is inserted after the complementizer as illustrated in (16f), the C-t effect does not occur. Adverbials are phonologically prominent; the high pitch that occurs on the complementizer continues to rise and the final pitch peak occurs on the following adverbial.¹² The C-t effect does not occur in right node raising (16g), either. The first intermediate phrase starts with *who* and ends with *whether*, and the second one starts with and and ends with the complementizer that, after which an intonational break occurs. At this point, the high pitch is suspended on the complementizer, and the third intermediate phrase starts with the verbal head *cheated*. The highest pitch peak of the third one should occur on cheated since the pitch must lower finally in the entire wh-question. Thus, no conflict of the pitch level arises on the verbal head following the complementizer.¹³

Though detailed phonological/intonational properties are not clear,¹⁴ the generalization here is expected to apply to Nupe, which shows the C-t effect: since the pitch is reset after the complementizer as Kandybowicz (2006) states, the pitch is expected to rise after the complementizer. In the same way as in English, the C-t effect is mitigated when a phonologically prominent element follows the complementizer as illustrated in (17a), and also when the complementizer is reduced as illustrated in (17b).

¹¹ Kandybowicz states that in matrix subject relative clauses and clefts, the complementizer must be overt:

i) the boy [*(that) bottles fireflies]

ii) It's my cousin [*(that) bottles fireflies].

This is purely a matter of English grammar: the overt complementizer cannot be omitted to construct subject relative clauses and cleft sentences.

¹² This is confirmed by my data, which is not presented here. In the following test sentence,

i) Who do you think [that under no circumstances would betray you]?

the final pitch peak occurs on the adverbial phrase under no circumstances.

¹³ Kandybowicz states that *that*-clausal elements cannot be focused except the verbal head:

i) [?]Who did you say that __ WROTE Barriers yesterday? ii)^{??/*}Who did you say that __ wrote BARRIERS yesterday?

iii) *Who did you say that ___ wrote Barriers YESTERDAY?

iv) *Who do you THINK that ____ wrote Barriers (as opposed to say, know)?

For one thing, a sentence cannot have two foci since one sentence cannot make two assertions (Lambrecht 1994). For the other thing, the pitch rises on a (contrastively) focused phrase. This contradicts the pitch gesture of whquestions in which the pitch falls after the highest peak on a wh-phrase and continues to fall. (ii-iv) is ruled out for these reasons. In the pitch gesture of the speakers who do not permit the C-t context, the pitch is reset and rises on the complementizer. When the pitch level on the verbal head that follows the complementizer is high, no conflict of the pitch level arises on that verbal head. Thus, the prominence on the verbal head that directly follows the complementizer is permitted.

¹⁴ See Hyman (2003) for the phonology of the African languages, and also Féry (2017) for an introduction of African lexical tones.

- (17) a. Zèé Musa gàn [gànán pányi lèé __ nì enyà] o? [Nupe]
 who Musa say that before PAST beat drum o
 'Who did Musa say that a long time ago beat the drum?'
 - b. Zèé Musa gàn ['án __ nì enyà] o?
 who Musa say that beat drum o
 'Who did Musa say th't beat the drum?'
 - c. [intP Zèé u: bè ke] [intP u: du nakàn na o]?
 who 3rd-sg seem that 3rd-sg cook meat na o
 'Who does it seem cooked the meat?'

The pitch rises on the complementizer and continues to rise up to the following element in (17a); the pitch lowers on the reduced complementizer and is low on the following nonprominent verbal head in (17b). Thus, no conflict of the pitch level arises on the element following the complementizer in either of the cases. Recall that the Nupe complementizer is aligned at the right edge of the intermediate phrase composed of main-clausal elements, i.e. ... ke][intP Interestingly, it seems that this alignment of the complementizer is strictly obeyed in Nupe, and the pitch is obligatorily reset after the complementizer. Kandybowicz states that regular phonological processes are blocked between the complementizer and the following element. In (17c), to avoid hiatus, glide formation could occur between *ke* 'that' and *u*: (person marker), which would result in [kju:]. But due to the strict alignment of the complementizer at the right edge of the intermediate phrase composed of main-clausal elements and the obligatory pitch resetting after the complementizer, glide formation is blocked.¹⁵

Several predictions are made based on generalization (14). First, in a language that has LH as its basic contour, the C-t effect is likely to occur. This case is illustrated by French.¹⁶ The extraction of a *wh*-object across the complementizer *que* is acceptable (18a) but that of a *wh*-subject is not (18b) (Rizzi and Shlonsky 2007).

(18) a. Qui crois-tu que Paul va aider ? [Fre.] who believe-you that Paul will help 'Who do you believe that Paul will help?'

¹⁵ Kandybowicz claims that cases such as (17c), in which a function word such as a resumptive pronoun, i.e. *u*: (or a tense marker in other cases) follows the overt complementizer, illustrate the mitigation of the C-t effect. But as stated above, those cases simply illustrate the blocking of glide formation due to the strict alignment of the complementizer and the obligatory pitch resetting, not the mitigation of the C-t effect.

¹⁶ For the French intonational system, see Di Cristo (1998), Jun and Fougeron (2000), Gussenhoven (2004) and Féry (2017). See also Deprez et al. (2013) for an interesting discussion on the prosody of the French *wh*-question.

b. *Qui crois-tu que va gagner?
who believe-you that will win
'Who do you believe will win?'

In French simple *wh*-questions such as *a qui as-tu prêté ce livre?* (to whom have-you lent this book 'to whom did you lend this book?'), the highest pitch peak occurs on the sentence-initial *wh*-phrase, and the pitch continues to fall until the end of the entire sentence (Di Cristo 1998, Jun and Fougeron 2000). According to Di Cristo (1998), the pronoun *tu* 'you' composes a prosodic unit with the preceding *crois* 'believe' as in the context of (18a). Since the rightmost full vowel is assigned a stress in such a prosodic unit, the pronoun is exceptionally stressed, which causes the following overt complementizer to be aligned at the left edge of the intermediate phrase composed of *que*-clausal elements, i.e. ... crois-tu][intP que It is expected that the pitch is reset on the complementizer and a high contour occurs after the complementizer. When a phonologically prominent element follows the complementizer as in (18a), the pitch gesture does not contradict the pitch movement of the basic LH contour. When a phonologically non-prominent verbal head follows the complementizer as in (18b), the pitch gesture contradicts the LH movement, which makes this sentence unacceptable.¹⁷

Secondly, in a language in which the pitch does not lower in *wh*-questions in the unmarked case, the C-t effect is likely to occur. This case is illustrated by Russian.¹⁸ The extraction of a *wh*-object is barely permitted as in (19a), and the extraction of a *wh*-subject is completely not permitted as in (19b) (Pesetsky 2017).

- (19) a. [%]Kogo ty xočeš', čtoby Maša vstretila __? [Rus.]
 who you want that Maša meet
 'Who do you want Masha to meet?'
 - b. *Kto ty xočeš', čtoby __ vstretil Mašu?
 who you want that meet Maša
 'Who do you want to meet Masha?'

[Fre.]

¹⁷ An alternative account for the ungrammaticality of (18b) is that contrary to (18a), in which the complementizer composes a prosodic unit with the following prominent *Paul*, the complementizer cannot make a prosodic unit with any element and is phonologically/intonationally isolated in the sentence. It is reported that when the *que* form is changed to *qui*, the acceptability slightly rises (Rizzi and Shlonsky 2007):

i) %Quelle étudiante crois-tu qui va partir?

which student believe you that will leave

^{&#}x27;Which student do you think will leave?'

The *que* form is composed of a consonant only, i.e. [k], whereas *qui* is composed of the consonant and a following vowel, i.e. [ki]. The *qui* form may manage to compose a prosodic unit by itself. I leave this possibility for future study.

¹⁸ See Svetozarova (1998) for the Russian intonational system.

In Russian *wh*-questions, the pitch rises on the initial *wh*-phrase and continues to be high until the end of the sentence, where the pitch finally falls (Svetozarova 1998). In such a language, a complex *wh*-question will be dispreferred: a high pitch cannot last so long. When a phonologically prominent element, the subject *Maša*, follows the complementizer as in (19a), the pitch is not low on the subject and finally falls on the verbal head *vstretila* 'meet'. Since the pitch gesture does not contradict the continuation of a high pitch, this sentence is barely permitted. When the verbal head immediately follows the complementizer as in (19b), the pitch should be low but actually rises on the non-prominent verbal head. The conflict of the pitch level occurs on that verbal head; this sentence is totally unacceptable.

5. The Optionality, Preference and Obligatoriness of an Overt Complementizer

This section presents phonetic data of the constructions relevant to the C-t effect in languages that do not show the C-t effect, and discusses the optionality, preference and obligatoriness of an overt complementizer. In Italian, both the extraction of a *wh*-subject and that of a *wh*-object across an overt complementizer are acceptable (Rizzi 1982). In (20a-b), the *wh*-subject *chi* 'who' is extracted; in (21a-b), the *wh*-object *cosa* 'what' is extracted. The complementizer *che* 'that' can occur optionally in both cases.

- (20) a. Chi pensi [che __ abbia scritto il libro]? [Ita.]
 b. Chi pensi [Ø __ abbia scritto il libro]? who you-think that has wirten the book 'Who do you think (that) wrote the book?'
- (21) a. Cosa pensi [che Giovanni abbia scritto]?
 b. Cosa pensi [Ø Giovanni abbia scritto]?
 what you-think that Giovanni has written
 'What do you think (that) Giovanni wrote?'

The pitch patterns of an Italian native speaker (female, born in Milan, Italy) are presented below:¹⁹

¹⁹ See Rossi (1998), D'Imperio and Rosenthall (1999) and Grice et al. (2005) for the Italian intonational system.



(22) a. Cosa pensi Giovanni abbia scritto? ('what do you think Giovanni wrote?')

b. Che cosa pensi che Giovanni abbia scritto? ('what do you think that Gio. wrote?')



In the extraction of a *wh*-object, the entire sentence is pronounced as one intonational phrase, whether the complementizer *che* 'that' is overt as in (22b) or not as in (22a). The pitch peak occurs on the sentence-initial *wh*-phrase. The pitch is not reset on the complementizer and continues to fall until the end of the entire sentence, but it slightly rises on the final word (cf. Rossi 1998). When the complementizer is overt as in (22b), an intonational break occurs after the overt complementizer *che* 'that'. The overt complementizer is aligned at the right edge of the intermediate phrase composed of main-clausal elements, i.e. ... che][_{intP}

The pitch patterns of the construction of *wh*-subject extraction are presented below:





In the extraction of a *wh*-subject too, the entire sentence is pronounced as one intonational phrase, whether the complementizer is overt as in (23b) or not as in (23a). The pitch peak occurs on the sentence-initial *wh*-phrase. The pitch is not reset on the complementizer and continues to fall until the end of the entire sentence, but it slightly rises on the final word. The overt complementizer *che* [ke] is combined with the first vowel *a*- of the following Aux *abbia* 'has', which produces a word-like unit, *che-abbia* [kjabia]; in the context of hiatus, a glide *j* is inserted. This contraction causes the complementizer *che* to be aligned at the left edge of the intermediate phrase composed of *che*-clausal elements, i.e. ...][intP che(-abbia) ..., with an intonational break present before the complementizer.

In the same way as in the pitch gesture of the English and Finnish native speakers who permit the C-t context, the pitch is not reset on the overt complementizer *che* 'that' and continues to fall until when it slightly rises on a final word. The final pitch peak does not occur within a complement clause. The pitch continues to fall through the complementizer to a following sentential element, whether the latter is a prominent nominal (22b) or a non-prominent verbal head/Aux (23b). No conflict of the pitch level arises on the element following the complementizer, and the C-t effect does not arise. Since most of the speakers show this pitch gesture, the presence of the overt complementizer is optional in Italian. The prosodic phrasing of the overt complementizer is also optional, which depends on the phonological condition under which the contraction between the complementizer and the following Aux occurs.

In Finland-Swedish, a variety of Swedish spoken in Finland, both the extraction of a *wh*-subject (24a-b) and that of a *wh*-object (25a-b) across the overt complementizer *att* 'that' are acceptable (Holmberg 1986). But the presence of the complementizer as illustrated in (24-25a) is preferred. That is, regardless of whether a phonologically prominent element is adjacent to the complementizer, the presence of the complementizer is preferable in Finland-Swedish.²⁰

²⁰ It should be noted that we here stick to Finland-Swedish. Most of the Swedish varieties spoken in Sweden show the C-t effect. Specifically, (24a) is ungrammatical for the speakers from Sweden.

- (24) a. Vem tror du [att __ skrev boken]?
 b. Vem tror du [Ø_ skrev boken]? who think you that wrote the book
 'Who do you think (that) wrote the book?'
- (25) a. Vad tror du [att Jon skrev]?
 b. Vad tror du [Ø Jon skrev]?
 what think you that Jon wrote
 'What do you think (that) Jon wrote?'

The pitch patterns of a Finland-Swedish native speaker (male, born in Turku, Finland) are presented:²¹



In the extraction of a wh-object, the entire sentence is pronounced as one intonational phrase, whether the complementizer *att* 'that' is overt as in (26b) or not as in (26a). The pitch peak occurs on the sentence-initial wh-phrase. The pitch is not reset on the complementizer and continues to fall until the end of the entire sentence. When the complementizer is overt as in

[Swe.]

²¹ See Bruce (1977, 2005, 2007), Gårding (1998), Gussenhoven (2004), Riad (2014) and Féry (2017) for the Swedish intonational system.

(26b), an intonational break occurs after the overt complementizer *att* 'that'. The overt complementizer is aligned at the right edge of the intermediate phrase composed of mainclausal elements, i.e. ... att][$_{intP}$

In the extraction of a *wh*-subject too, the entire sentence is pronounced as one intonational phrase, whether the complementizer is overt as in (27b) or not as in (27a). The pitch peak occurs on the sentence-initial *wh*-phrase. The pitch is not reset on the complementizer and continues to fall until the end of the entire sentence.



(27) a. Vem tror du skrev boken? ('who do you think wrote the book?')

b. Vem tror du att skrev boken? ('who do you think that wrote the book?')



In the same way as in the pitch gesture of those who permit the C-t context, after the pitch peak occurs on the sentence-initial *wh*-phrase, the pitch falls until the end of the entire sentence. In Finland-Swedish, a focus is manifested by a high pitch peak (Bruce 2005, 2007). Since the *wh*-phrases *vad* 'what' and *vem* 'who' are monosyllabic words, the high pitch that occurs on them continues up to the following verbal head *tror* 'think', on which the pitch falls. When the overt complementizer *att* 'that' is absent as illustrated in (26-27a), the pitch is still high on the subject *du* 'you'; the pitch will not easily lower on the element that immediately follows *du*. When the overt complementizer is present as illustrated in (26-27b), it is aligned at the right edge of the intermediate phrase composed of main-clausal elements, i.e. ... att][intP The pitch continues to fall until *att* 'that', after which an intonational break occurs. The low

pitch on the complementizer is kept on the following element, and the pitch is likely to lower on the latter, which accounts for the preference of the overt complementizer in Finland-Swedish.²²

Dutch obligatorily requires an overt complementizer, whether a *wh*-subject or any other sentential element is extracted (Perlmutter 1971). In (28a-b), the *wh*-subject *wie* 'who' is extracted; in (29a-b), the *wh*-object *wat* 'what' is extracted. Dutch is an SOV language in that the word order of embedded clauses is SOV: the arguments of a verb as well as adverbials are placed before a main verb in embedded clauses.²³ Regardless of whether a *wh*-subject or a *wh*-object is extracted, a phonologically prominent element is adjacent to the complementizer: in (28a-b), the object *het boek* 'the book' is adjacent to the complementizer; in (29a-b), the subject *Bill* is adjacent to the complementizer. Thus, the presence of the complementizer is obligatory in Dutch even when a phonologically prominent element is adjacent to it.

- (28) a. Wie denk je [dat __ het boek heeft geschreven]? [Dut.]
 b. *Wie denk je [Ø __ het boek heeft geschreven]? who think you that the book has written
 'Who do you think (that) wrote the book?'
- (29) a. Wat denk je [dat Bill __ heeft geschreven]?
 b. *Wat denk je [Ø Bill __ heeft geschreven]? what think you that Bill has written 'What do you think that Bill wrote?'

The pitch patterns of a Dutch native speaker (male, born in Amsterdam, The Netherlands) are presented below.²⁴ In the extraction of a *wh*-object as illustrated in (30), the entire sentence is pronounced as one intonational phrase. The pitch peak occurs on the sentence-initial *wh*-phrase. The pitch is not reset on the complementizer and continues to fall until the end of the entire sentence. An intonational break occurs before the complementizer *dat* 'that'. The overt complementizer is aligned at the left edge of the intermediate phrase composed of *dat*-clausal elements, i.e.][intP dat

²² As stated in footnote 20, the Swedish speakers from Sweden do show the C-t effect. The properties of the pitch gesture of the Swedish varieties in Sweden differ from those of Finland-Swedish (Bruce 2005, 2007). I leave a detailed phonetic analysis on the C-t effect in the entire Swedish varieties for future.

²³ In main clauses too, the word order is SOV when a sentence has an Aux: Subj Aux ... VPP-main.

²⁴ See Gussenhoven and Rietveld (1988, 2000), t' Hart et al. (1990), Sluijter and van Heaven (1995), Grabe et al. (1997), 't Hart (1998) and Kirsner et al. (1998) for the Dutch intonational system.



In the extraction of a *wh*-subject too, the entire sentence is pronounced as one intonational phrase. The pitch peak occurs on the sentence-initial *wh*-phrase. The pitch is not reset on the complementizer and continues to fall until the end of the entire sentence; see (31a). The absence of the complementizer is actually acceptable in the extraction of a *wh*-subject. But it is necessary to change the word order by fronting the Aux *heeft* 'has' before the object *het boek* 'the book', which construction is pronounced as one intonational phrase; see (31b).²⁵



b. Wie denk je heeft het boek geschreven? ('who do you think wrote the book?')



²⁵ In my ongoing research, it has turmed out that this sentence pattern is acceptable only when *denk je* is interpreted as an inserted phrase, i.e. *Who, do you think, wrote the book?*. I leave a more precise phonetic analysis of this construction for future.

In the same way as in the pitch gesture of the speakers who permit the C-t context, after the pitch peak occurs on the sentence-initial *wh*-phrase, the pitch falls until the end of the entire sentence. In SOV languages such as Dutch, the highest prominence is likely to be assigned to the position right before a main verb (Gundel 1988). Especially, Dutch has the *hat pattern* ('t Hart 1998), in which the pitch rises on a prominent word, and a high pitch continues until when the pitch falls on the last accentable syllable. Thus, the pitch could rise on a prominent element of an embedded clause, whether it is the subject *Bill* (30) or the object *het boek* 'the book' (31a). When the overt complementizer *dat* 'that' is inserted, an intonational break occurs before *dat*. It is aligned at the left edge of the intermediate phrase composed of *dat*-clausal elements, i.e.][intP dat The pitch is not reset on the complementizer, and the pitch level on it is kept on the following element, which enables the pitch to lower smoothly. Thus, the presence of the overt complementizer is near obligatory in Dutch.

In sum, the optionality, preference and obligatoriness of an overt complementizer depends on different phonological/intonational environments in different languages, though the lowering pitch gesture of *wh*-questions does not differ. In the environment where nothing prevents the pitch from lowering, the insertion of an overt complementizer is optional, as illustrated by Italian. In the environment where the pitch is more difficult to lower, the insertion of an overt complementizer is more preferable as in Swedish and even obligatory as in Dutch. The inserted complementizer acts as keeping the pitch level and enables the pitch to lower smoothly.

6. Conclusion

I have presented a generalization on the C-t effect from the intonational perspective on the basis of a comparative investigation collecting phonetic data from English and Finnish, in both of which the C-t context is acceptable to some speakers but unacceptable to others, as well as from Italian, Swedish and Dutch, in which the presence of an overt complementizer is either optional (Ita.), preferable (Swe.) or obligatory (Dut.). The generalization on the C-t effect based on the data from English and Finnish is that in the pitch gesture of the speaker who shows the C-t effect, the pitch is reset on the overt complementizer and the final pitch accent occurs within the complementizer clause; in the pitch gesture of the speaker who permits the C-t context, the pitch is not reset on the overt complementizer and continues to lower. This generalization applies to an individual speaker, not to an individual language; the more native speakers of a language it applies to, that language is more likely to show the C-t effect. It has thus been argued that the generalization here accounts for why the acceptability of the C-t context differs between languages as well as between the native speakers of a language. The C-t effect has been accounted for from the intonational perspective in terms of the conflict of the pitch level on the

element following the overt complementizer, i.e., the pitch should lower but actually rises on it. Based on the comparative study between Italian, Finland-Swedish and Dutch, all of which do not show the C-t effect, the optionality, preference and obligatoriness of an overt complementizer has been discussed. It has been argued that in the phonological/intonational environment where the pitch is more difficult to lower, the insertion of an overt complementizer is more preferable, and that the inserted complementizer acts as keeping the pitch level and enables the pitch to lower smoothly.

A more interesting generalization on the C-t effect from the intonational perspective is expected to be gained in the future study. According to Pesetsky (2017), Wolof shows the C-t effect. Both Wolof and Nupe are Niger-Congo African. African languages are tone languages and have downstep which is caused by some lowering trigger. It is quite interesting if the C-t effect arises only in Niger-Congo varieties among all African languages. Pesetsky also states that some of the Arabic varieties show the C-t effect. I have collected data from speakers of some Arabic varieties. The data is so complicated that I leave the analysis on the C-t effect in Arabic for future. According to Maling and Zaenen (1978), Icelandic obligatorily needs an overt complementizer. With the facts on Finland-Swedish introduced in this paper taken into account, an interesting micro-parametric difference on the C-t effect between the Scandinavian languages from the intonational perspective will be revealed. Interestingly, Featherston (2005) reports that in German, which is a SOV language in the same way as Dutch, both the extraction of a *wh*-subject and that of a *wh*-object across an overt complementizer are degraded. It is highly expected that when an overt complementizer appears, the pitch is reset on it, which I leave for future study.

References

- Boersma, Paul, and David Weenink. 1996. Praat, a system for doing phonetics by computer, version 3.4. *Institute of Phonetic Sciences of the University of Amsterdam, Report* 132.
- Bolinger, Dwight. 1978. Intonation across languages. In *Universals of Human Language Vol.2: Phonology*, ed. by Joseph H. Greenberg, 471-524. Stanford, CA.: Stanford University Press.
- Bolinger, Dwight. 1998. Intonation in American English. In *Intonation Systems: A Survey of Twenty Languages*, ed. by Daniel Hirst, and Albert Di Cristo, 45-55. Cambridge: Cambridge University Press.
- Bošković, Željko. 2011. Rescue by PF deletion, traces as (non)interveners, and the that-trace effect. *Linguistic Inquiry* 42:1-44.
- Bresnan, Joan. 1977. Variables in the Theory of Transformations. In *Formal Syntax*, ed. by Peter Culicover, Thomas Wasow and Adrien Akmajian, 157-196. New York: Academic Press.
- Bruce, Gösta. 1977. Swedish Word accents in sentence perspective. *Travaux de L'Institut de Linguistique de Lund* XII. CWK GLEERUP.
- Bruce, Gösta. 2005. Intonational Prominence in Varieties of Swedish Revisited. In Prosodic

Typology: The Phonology of Intonation and Phrasing, ed. by Sun-Ah Jun, 410-429. Oxford: Oxford University Press.

Bruce, Gösta. 2007. Components of a prosodic typology of Swedish intonation. In *Tones and Tunes Volume 1: Typological Studies in Word and Sentence Prosody*, ed. by Tomas Riad, and Carlos Gussenhoven, 113-146. Berlin: Mouton de Gruyter.

Chomsky, Noam. 1981. Lectures on Government and Binding. New York: Mouton de Gruyter.

- Chomsky, Noam. 1986. Knowledge of Language. Westport, CT: Praeger.
- Chomsky, Noam. 1995. The Minimalist Program. Cambridge, MA: MIT Press.
- Chomsky, Noam. 2001. Derivation by phase. In *Ken Hale: A Life in Language*, ed. by Michael Kenstowicz, 1-52. Cambridge, MA: MIT Press.
- Chomsky, Noam. 2008. On phases. In *Foundational Issues in Linguistic Inquiry: Essays in Honor of Jean-Roger Vergnaud*, ed. by Robert Freidin, Carlos P. Otero, and Maria Luisa Zubizarreta, 133-166. Cambridge, MA: MIT Press.
- Chomsky, Noam. 2015. Problems of projection: Extensions. In Structures, Strategies and Beyond: Studies in Honour of Adriana Belletti, ed. by E. D. Domenico, C. Hamann and S. Matteini, 3-16. Amsterdam: John Benjamins.
- Chomsky, Noam, and Morris Halle. 1968. *The Sound Pattern of English*. Cambridge, MA: MIT Press.
- Cinque, Guglielmo. 1993. A Null Theory of Phrase and Compound Stress. *Linguistic Inquiry* 24:239-297.
- Cowart, Wayne. 1997. Experimental Syntax: Applying Objective Methods to Sentence Judgments. Thousand Oaks, CA: Sage Publications.
- Cowart, Wayne. 2003. Detecting syntactic dialects: The *that*-trace phenomenon. Talk delivered at 39th Regional Meeting of the Chicago Linguistic Society.
- Deprez, Viviane, Kristen Syrett, and Shigeto Kawahara. 2013. The interaction of syntax, prosody, and discourse in licensing French *wh-in-situ* questions. *Lingua* 124:4-19.
- Di Cristo, Albert. 1998. Intonation in French. In *Intonation Systems: A Survey of Twenty Languages*, ed. by Daniel Hirst, and Albert Di Cristo, 195-218. Cambridge: Cambridge University Press.
- D'Imperio, Mariapaola, and Sam Rosenthall. 1999. Phonetics and Phonology of Main Stress in Italian. *Phonology* 16:1-28.
- Featherston, Sam. 2005. That-trace in German. Lingua 115:1277-1302.
- Féry, Caroline. 2017. *Intonation and Prosodic Structure*. Cambridge: Cambridge University Press.
- Gårding, Eva. 1998. Intonation in Swedish. In *Intonation Systems: A Survey of Twenty Languages*, ed. by Daniel Hirst, and Albert Di Cristo, 112-130. Cambridge: Cambridge University Press.
- Gordon, Matthew.K. 2016. Phonological Typology. Oxford: Oxford University Press.
- Grabe, Esther, Carlos Gussenhoven, Judith Haan, Erwin Marsi and Brechtje Post. 1997. Preaccentual pitch and speaker attitude in Dutch. *Language and Speech* 41:63-85.

Grice, Martine, Mariapaola D'Imperio, Michelina Savino and Cinzia Avesani. 2005. Strategies for intonation labelling across varieties of Italian. In *Prosodic Typology: The Phonology of Intonation and Phrasing*, ed. by Sun-Ah Jun, 362-389. Oxford: Oxford University Press.

Gundel, Jeanette K. 1988. Universals of topic-comment structure. In Studies in Syntactic

Typology, ed. by Michael Hammond, Edith Moravcsik, and Jessica Wirth, 209-239. Amsterdam: John Benjamins Publishing Company.

- Gussenhoven, Carlos. 2004. *The Phonology of Tone and Intonation*. Cambridge: Cambridge University Press.
- Gussenhoven, Carlos, and Toni Rietveld. 1988. Fundamental frequency declination in Dutch: testing three hypotheses. *Journal of Phonetics* 16:355-369.
- Gussenhoven, Carlos, and Toni Rietveld. 2000. The behavior of H *and L* under variations in pitch range in Dutch rising contours. *Language and Speech* 43:183-203.
- 't Hart, Johan. 1998. Intonation in Dutch. In *Intonation Systems: A Survey of Twenty Languages*, ed. by Daniel Hirst, and Albert Di Cristo, 96-111. Cambridge: Cambridge University Press.
- 't Hart, Johan, René Collier, and Antonie Cohen. 1990. *A Perceptual Study of Intonation*. Cambridge: Cambridge University Press.
- Hirst, Daniel. 1998. Intonation in British English. In *Intonation Systems: A Survey of Twenty Languages*, ed. by Daniel Hirst, and Albert Di Cristo, 56-77. Cambridge: Cambridge University Press.
- Holmberg, Anders. 1986. Word order and syntactic features in the Scandinavian languages and English. PhD dissertation, University of Stockholm.
- Hyman, Larry M. 2003. African languages and phonological theory. *GLOT International* Vol. 7(6):153-163.
- Iivonen, Antti. 1998. Intonation in Finnish. In Intonation Systems: A Survey of Twenty Languages, ed. by Daniel Hirst, and Albert Di Cristo, 314-330. Cambridge: Cambridge University Press.
- Jun, Sun-Ah, and Cécile Fougeron. 2000. A Phonological model of French intonation. In Intonation: Analysis, Modeling and Technology, ed. by Antonis Botinis, 209-242. Dordrecht: Kluwer Academic Publishers.
- Kandybowicz, Jason. 2006. Comp-trace effects explained away. In *Proceedings of the 25th West Coast Conference on Formal Linguistics*, ed. by Donald Baumer, David Montero, and Michael Scanlon, 220-228.
- Kirsner, Robert S., Vincent van Heuven and Johanneke Caspers. 1998. From request to command: An exploratory experimental study of grammatical form, intonation, and pragmatic particle in Dutch imperatives. *Linguistics in the Netherlands*, 135-148.
- Ladd, D. Robert. 2008. Intonational Phonology. Cambridge: Cambridge University Press.
- Lambrecht, Knud. 1994. Information Structure and Sentence Form: Topic, Focus, and the Mental Representations of Discourse Referents. Cambridge: Cambridge University Press.
- Maling, Joan, and Annie Zaenen. 1978. The nonuniversality of a surface filter. *Linguistic Inquiry* 9:475-497.
- McFadden, Thomas, and Sandhya Sundaresan. 2018. What the EPP and comp-trace effects have in common: constraining silent elements at the edge. *Glossa: A Journal of General Linguistics* 43:1-34.
- Nakai, Satsuki, Sari Kunnari, Alice Turk, Kari Suomi and Riikka Ylitalo. 2009. Utterance-final lengthening and quantity in Northern Finnish. *Journal of Phonetics* 37:29-45.
- Perlmutter, David M. 1971. *Deep and Surface Structure Constraints in Syntax*. New York: Holt, Rinehart and Winston.
- Pesetsky, David. 2017. Complementizer-trace effects. In The Wiley Blackwell Companion to

Syntax, 2nd edition, ed. by Martin Everaert and Henk C. Van Riemsdijk. Oxford: Wiley-Blackwell.

- Pierrehumbert, Janet. 1980. The phonology and phonetics of English intonation. PhD dissertation. MIT (published by Garland Press, New York, 1990).
- Riad, Tomas. 2014. The Phonology of Swedish. Oxford: Oxford University Press.
- Richards, Marc. 2007. On feature inheritance: An argument from the phase impenetrability condition. *Linguistic Inquiry* 38: 563-572.
- Ritchart, Amanda, Grant Goodall and Marc Garellek. 2016. Prosody and the that-trace effect: An experimental study. *Proceedings of the 33rd West Coast Conference on Formal Linguistics*, 320-328.
- Rizzi, Luigi. 1982. Issues in Italian Syntax. Dordrecht: Foris.
- Rizzi, Luigi, and Ur Shlonsky. 2007. Strategies of subject extraction. In *Interfaces + Recursion* = *Language?: Chomsky's Minimalism and the View from Syntax-Semantics*, ed. by Uli Sauerland and Hans-Martin Gärtner, 115-160. Berlin: Mouton de Gruyter.
- Rossi, Mario. 1998. Intonation in Italian. In *Intonation Systems: A Survey of Twenty Languages*, ed. by Daniel Hirst, and Albert Di Cristo, 219-241. Cambridge: Cambridge University Press.
- Selkirk, Elisabeth O. 1984. *Phonology and Syntax: The Relation between Sound and Structure*. Cambridge, MA: MIT Press.
- Selkirk, Elisabeth. 1995. Sentence prosody: Intonation, stress, and phrasing. In *The Handbook* of *Phonological Theory*, ed. by John A. Goldsmith, 550-569. London: Blackwell.
- Sluijter, Agaath M.C., and Vincent van Heuven. 1995. Effects of focus distribution, pitch accent and lexical stress on the temporal organization of syllables in Dutch. *Phonetica* 52:71-89.
- Suomi, Kari, Juhani Toivanen and Riikka Ylitalo. 2008. *Finnish Sound Structure: Phonetics, Phonology, Phonotactics and Prosody*. Oulu: Oulu University Press.
- Svetozarova, Natalia. 1998. Intonation in Russian. In *Intonation Systems: A Survey of Twenty Languages*, ed. by Daniel Hirst, and Albert Di Cristo, 264-277. Cambridge: Cambridge University Press.

Mayumi Hosono Keio University mayumi.hososo@keio.jp

On the relative order of central sentence adverbs in the Insular Scandinavian languages

Ásgrímur Angantýsson, University of Iceland

Abstract

This paper discusses the relative order of certain classes of central sentence adverbs in Icelandic and Faroese. The relative order of the logical subject and central sentence adverbs in double subject constructions is also taken under consideration. The questionnaire data shows that the relative orders of adverbs that follow Cinque's (1999) hierarchy receive more positive judgments than the orders that do not exhibit the predicted patterns in both languages, but the contrasts are not always as striking as one might expect. Examples of three adverbs within the same clause get rather negative judgments in general, but in case such orders are accepted the judgments are also in accordance with Cinque's (1999) hierarchy. In double subject constructions, however, the sentence adverbs usually must precede the logical subject in Faroese while in Icelandic both orders are fine for most speakers. This is consistent with recent research showing that adverb placement is somewhat less free in Faroese than in Icelandic.

1. Introduction¹

This paper discusses the relative order of central sentence adverbs in Icelandic and Faroese, specifically the so-called speech act adverbs and evidential adverbs as in (1), and conjunctive adverbs and evaluative adverbs as in (2). The relative order of the logical subject and central sentence adverbs in double subject constructions as in (3) is also taken under consideration (see discussions on such orders in Icelandic in Jónsson 2002).

(1)		ur satt at siga honestly	obviously	made a mistake		
		< 1	(speech act \rightarrow evidentiality)			
	b. Jón hev		iga satt at siga gjørt eitt mistak.			
		(evidentiality	(evidentiality \rightarrow speech act)			
(2)	a. Hanus v	var tó	<u>tíbetur</u> slo	oppin óskaddur.		
	Hanus l	nad though	thankfully escaped unharmed			
		(conjunction -	$(conjunction \rightarrow evaluation)$			
	b. Hanus v	var <u>tíbetur</u>	tó sloppi	n óskaddur.		
		(evaluation	(evaluation \rightarrow conjunction)			
(3)	a. Tað ha	ava tíbetur	<u>nógv</u> lisið bók	tina.		
	Expl. h	ave fortunately	fortunately many read the book			
	-	(evaluation \rightarrow	(evaluation $\rightarrow \log$. subject)			
	b. Tað ha	ava <u>nógv</u>	tíbetur lisið bo	ókina.		
		-	(log. subject \rightarrow evaluation)			
	'Many I		ve fortunately read the book'			

¹ I want to thank Johan Brandtler for useful comments and corrections. I am also indebted to Höskuldur Thráinsson and two anonymous reviewers from *Íslenskt mál* for their helpful comments on an earlier (Icelandic) version of this paper (Angantýsson 2017).

The aim is on the one hand to show both similarities and differences in this regard between the two related languages and on the other hand to connect the results with academic theories about the relation of adverbs to other words and parts of sentences.

Cinque (1999) proposed a famous theory on the order of adverbs and adverbial phrases in the world's languages. To simplify, the word "adverb" will be used in this paper, whether it refers to a single adverb or adverbial phrases that form a semantic whole. According to Cinque's theory, the order of adverbs is determined by meaning and the word order restrictions are described as a specific hierarchy (see chapter 2). The data introduced here, show that the relative orders of central adverbs that are in accordance with Cinque's hierarchy are considered far better in both languages than the orders that are not in accordance with the hierarchy. In that regard for example, far more Faroese speakers accept (1a) and (2a) than (1b) and (2b) but the contrasts are actually not as striking as one might expect from the theory on hierarchy. In some cases, the restraints on word order seem to be more rigid in Faroese than in Icelandic. This finding is in accordance with previous research that indicate that rules on word order are to some extent firmer in Faroese (see Angantýsson 2018 and references included there).

The layout is as following. Chapter 2 discusses the relevant adverbs and briefly outlines the structural ideas. Chapter 3 outlines and discusses results from questionnaire surveys that the author conducted in Iceland in 2015 and 2017 and in the Faroe Islands in 2016. Chapter 4 contains the conclusion.

2. Theoretical background

All languages contain some form of adverbs that usually connect with the main verb in the sentence in a sematic way and describe events: *The guitar player played well*. Among the points that linguists have discussed are the following (see general discussion about adverbs in Jackendoff 1972, Travis 1988, Alexiadou 1994, 1997, Cinque 1999, Nielsen 2000, Ernst 2002, 2004, 2007, Svenonius 2002, Pittner et al (ed.) 2015; a discussion about Icelandic adverbs can be found for example in Bergsveinsson 1969, Jónsson 2002, Jóhannsdóttir 2005 and Thráinsson 2005:123–137 and 2007:37–40, 79–87):

- (4) a. Different semantic classes of adverbs have a tendency to be placed in different positions in sentences.
 - b. Syntactic analyses often assume that certain adverbs have a set place in the syntactic structure and the adverbs as such are often used to argue for the placement of other words and parts.
 - c. The same adverb can carry a different meaning or scope of meaning depending on its syntactic position.

- d. The syntactic characteristics of adverbs can be quite different depending on languages and dialects.
- c. Opinions vary on the way adverbs interact with other words and phrases and how they integrate into the syntactic structure, that is, if they are adjuncts or placed in the specifier position of particular functional projections.

The examples in (5) show the type of adverbs that will mainly be discussed in this paper:

(5) Speaker-oriented central sentence adverbs

- a. Maturinn er í hreinskilni sagt ekki nógu góður. The food is honestly not good enough.
 b. Þetta er sem betur fer að verða búið.
- This is fortunately almost over.
- c. Hann er **skiljanlega** miður sín. He is understandably devestated.
- d. Hún er sannarlega vel að sigrinum komin.She is certainly deservant of the victory.
- e. Jón hefur **líklega** aldrei lesið Njálu. John has probably never read Njála.

Adverbial phrases of this kind naturally follow immediately after the inflected verb (see overview of the classification of adverbs in Icelandic in Thráinsson 2007:37–40). If two or more adverbial phrases of this kind are placed together in a sentence, Cinque's theory (1999:106) predicts the following relative order:

- (6) The hierarchy of central adverbs
- a. Speech act adverbs

frankly, briefly, honestly

- b. **Evaluative adverbs** *fortunately, understandably, luckily*
- c. Evidential adverbs supposedly, apparently, truly
- d. **Modal adverbs** *arguably, necessarily, probably*

Cinque (1999) does not explicitly mention that a break in the hierarchy will lead to unacceptable sentences but there is no doubt in his presentation of the material that sentences that are not in accordance with the hierarchy are always considered less acceptable, either being labelled 'ungrammatical' (star) or 'deviant' (question mark).

Jónsson (2002) discusses the relative order of adverbial phrases like (6) in Icelandic and an additional category that Cinque (1999) does not include:

(7) Conjunctive adverbs

lastly, firstly, consequently

Jónsson's conclusion can be summed up by saying that word orders that follow Cinque's hierarchy (1999) are better than those that do not, but that the contrasts are not as striking as one might expect. As we will see, my conclusions indicate a similar pattern, both in terms of Icelandic and Faroese. Jónsson also claims that conjunctive adverbs like those in (7) can be placed either before or after evidential and evaluative adverbs but my data shows that most speakers are more comfortable with having the conjunctive adverbs precede such adverbs.

There are two major approaches to how adverbs are integrated into clause structure (see an overview in Alexiadou 2004; Delfitto 2006 and Pittner et.al. (ed.) 2015). First, there is the adjunction analysis under which adverbs are assumed to adjoin rather freely to any maximal projection and the result is acceptable as long as the adverb in question can receive proper interpretation (cf. Ernst 2002, 2004, 2007). Second, there is the hierarchical account of adverbs where the assumption is that adverbs occupy fixed positions in the syntactic structure (Alexiadou 1997; Cinque 1999; Cinque 2004).

In addition to the aforementioned variations on word orders, Jónsson (2002) discusses the flexible order of logical subjects and sentence adverbs in double subject construction:

(8) a. Það hafa **sennilega** <u>flestir</u> lesið bókina.

EXPL have probably most read book-the.

b. Það hafa <u>flestir</u> s**ennilega** lesið bókina. EXPL have most probably read book-the.

Jónsson claims that this flexibility is expected under an adjunction analysis of adverbs while under Cinque's theory one must stipulate a functional structure with multiple subject positions situated among the various adverbrelated positions. This discussion will not systematically compare these analyses, but review the data with regard to Cinque's theories (1999).
3. Data from speakers' questionnaires

3.1 About the data

The data introduced here is based on 1) an electronic web based questionnaire given to 30 students in the University of Akureyri in 2015, 2) a written questionnaire given to 32 students in Fróðskaparsetur Føroya in 2016 and 3) a written questionnaire given to 37 students in the University of Iceland in 2017. The University of Akureyri questionnaire contained 162 sentence examples that were all in some way connected to varying positions of adverbs. The questionnaire given to students in the University of Iceland contained 24 sentence examples that almost all addressed the relative order of adverbs. The Faroese questionnaire contained 105 sentences, 40 of which concerned differences in the relative order of adverbs. There were three possible responses available to each sentence in every questionnaire (this sample is from the Faroese questionnaire):

(9)	Ja = Góður setningur. Soleiðis kundi eg væl sagt.
	Good sentence, I could easily say that
	? = Ivasamur setningur. Eg kundi møguliga sagt
	Questionable sentence, I might say that

Nei = Ómøguligur setningur. Soleiðis kundi eg ikki sagt. Unacceptable sentence, I could not say that

SO.

In the discussions of possible relative order of adverbs that I know of, linguists have mainly used their own judgements. It can however prove difficult to assess delicate nuances such as these and when academic theories and "interests" are at stake, there is always a chance of partiality in data. Therefore, it is my opinion that it is a safer practice to gather different assessments when the intent is to make assumptions about the quality of certain word orders in individual languages. It is also interesting to see the differences and similarities in these matters in languages as closely related as Icelandic and Faroese.

3.2 Icelandic

As mentioned with regard to (6) before, Cinque's theory on adverbs (1999) assumes that the first examples in the following sentence pairs are better than the latter. This seems to be a correct assumption in the Icelandic questionnaire (UI):

	_ 37 speakers			
	Yes	?	No	Order
 (10) Jón hefur satt að segja greinilega gert mistök. Jón has honestly obviously made a mistake. 	76%	21%	3%	Speech act → evidentiality
 (11) Jón hefur <u>greinilega</u> satt að segja gert mistök. Jón has obviously honestly made a mistake. 	3%	32%	65%	evidentiality → speech act
 (12) Jón hefur satt að segja skiljanlega engan áhuga. Jón has honestly understandably no interest. 	30%	16%	54%	Speech act \rightarrow evaluation
 (13) Jón hefur <u>skiljanlega</u> satt að segja engan áhuga. Jón has understandably honestly no interest. 	9%	32%	59%	evaluation \rightarrow speech act

 Table 1 Relative order of two adverbs that convey speech acts, evidentiality and evaluation

The difference is most obvious in (10) and (11) where most speakers consider it better to place the speech act adverb before the evidential adverb. It might however seem surprising how badly received the examples (11-13) in *Table* I are. That includes example (12) which has the "right" relative order of adverbs according to Cinque (1999) but the corresponding example with a changed order (13) is considered even worse by the speakers.

There is a considerable difference in the perception of speakers of the relative order of conjunctive adverbs on one hand and evidential and evaluative adverbs on the other hand as visible in *Table 2*:

Table 2 Relative order of two adverbs that convey conjunction, evaluation and evidentiality.

	37 speakers			
	Yes	?	No	Order
(14) Hann hefur samt <u>greinilega</u> megrast.He has however obviously lost weight.	50%	25%	25%	conjunction \rightarrow evidentiality
(15) Hann hefur <u>greinilega</u> samt megrast. He has obviously however lost weight.	8%	22%	70%	evidentiality \rightarrow conjunction
 (16) Hann hafði samt <u>sem betur fer</u> sloppið ómeiddur. He had however fortunately escaped unhurt. 	65%	14%	21%	conjunction \rightarrow evaluation
 (17) Hann hafði <u>sem betur fer</u> samt sloppið ómeiddur. He had fortunately however escaped unhurt. 	30%	30%	40%	evaluation \rightarrow conjunction

Jónsson (2002) claims that the relative order of conjunctive adverbs with these types of adverbs is free in Icelandic but these results indicate that speakers are more inclined to place the conjunctive adverb before the others.²

Table 3 gives examples of different relative orders of three adverbs in the same sentence. Overall, these types of sentences receive negative judgements (UI):

	37 speakers			
	Yes	?	No	Order
 (18) María hefur satt að segja sem betur fer greinilega lesið bókina. María has honestly fortunately obviously read the book. 	29%	14%	57%	speech act \rightarrow evaluation \rightarrow evidentiality
(19) María hefur satt að segja greinilega sem betur fer lesið bókina. María has honestly obviously fortunately read the book.	21%	24%	55%	speech act \rightarrow evidentiality \rightarrow evaluation

 Table 3 Relative order of three adverbs that convey a speech act, evaluation and evidentiality

The difference is minimal here and in no way significant. To facilitate an easier assessment of this type of sentences it would likely help to place the sentences in the context of some kind of discourse, which was not the case in these questionnaires. As a result, the speakers might have found it far-fetched to imagine a situation where it would be considered normal to use so many adverbial phrases within the same sentence. As we will see later in the discussion the results of the Faroese questionnaire were however more decisive in this regard. Table 4 shows the relative order of the logical subject on one hand and evaluative, speech act and evidential adverbs on the other, in double subject constructions with a postponed subject. Jónsson (2002) claims that both orders are viable but that it is generally considered better to place a sentence adverb of this kind before the noun phrase. The assessments of the Icelandic speakers indicate that he is right (UA):

² As pointed out by Höskuldur Thráinsson, it would be natural to assume beforehand that the "weight" of adverbial phrases could impact their prime placement within a sentence and it is often said that lighter phrases are placed relatively early in a sentence while heavier once are placed later, which in some cases might have something to do with their semantic qualities. This is something worth pursuing with more research but as it stands this type of impact cannot be detected in the sentences tested.

	30 speakers			
	Yes	?	No	Order
(20) Það hafa satt að segja <u>margir</u> lesið bókina. EXPL have honestly many read the book.	60%	23%	17%	Speech act adverb → subject
 (21) Það hafa <u>margir</u> satt að segja lesið bókina. EXPL have many honestly read the book. 	43%	20%	37%	subject → speech act adverb
 (22) Það hafa sem betur fer <u>margir</u> lesið bókina. EXPL have fortunately many read the book. 	80%	11%	9%	evaluative adverb → subject
 (23) Það hafa <u>margir</u> sem betur fer lesið bókina. EXPL have many fortunately read the book. 	51%	14%	34%	subject → evaluative adverb
(24) Það hafa greinilega <u>margir</u> lesið bókina. EXPL have obviously many read the book.	97%	3%	0	evidential adverb →subject
(25) Það hafa <u>margir</u> greinilega lesið bókina. EXPL have many obviously read the book.	54%	6%	40%	subject → evidential adverb

Table 4 The relative order of a sentence adverb and a logical subject

Placing the sentence adverb before the subject is very well received in (22) and (24) but rather less so in (20). Corresponding sentences that place the subject before the sentence adverb (21, 23, 25) are less popular but are in no way deemed impossible. As pointed out by Jónsson (2002:79), the subject can convey a meaning of parts ('many from a certain group') or a general mass meaning ('many overall') in examples such as those in *Table 4* and regardless of word order. If we assume a flexible position of adverbs this is not surprising but according to Cinque's ideas of structure (1999) we would have to assume varying positions of the subject in examples such as (20) and (21) even though the meaning would be the same (see discussion in Jónsson 2002).

3.3 Faroese

Let us now look at comparable data from Faroese. *Table 5* contains sentence pairs where the first example reflects the order expected according to Cinque's theories (1999):

	32 speakers				
	Yes	?	No	Order	
 (26) Jón hevur satt at siga <u>týðiliga</u> gjørt eitt mistak. John has honestly obviously made a mistake. 	73%	17%	10%	speech act \rightarrow evidentiality	
(27) Jón hevur <u>týðiliga</u> satt at siga gjørt eitt mistak. John has obviously honestly made a mistake.	10%	14%	76%	evidentiality → speech act	
(28) Jón hevur satt at siga <u>væl skiljandi</u> ongan áhuga. John has honestly understandably no interest.	42%	34%	24%	speech act \rightarrow evaluation	
(29) Jón hevur <u>væl skiljandi</u> satt at siga ongan áhuga. John has understandably honestly no interest.	38%	38%	24%	evaluation \rightarrow speech act	

 Table 5 The relative order of two adverbs that convey a speech act, evidentiality and evaluation

The difference in these sentence pairs is broadly similar to the corresponding examples from the Icelandic questionnaire. Cinque's order is far better received in the former pairing, but the results are almost the same in the latter and neither variation is received very well actually.

Table 6 shows the relative order of a conjunctive adverb with speech act adverbs and evidential adverbs in Faroese:

evidentiality		-	-	
	32 speakers			
	Yes	?	No	Order
 (30) Oddrún er tó <u>týðiliga</u> klænkað. Oddrún has though obviously lost weight. 	87%	3%	10%	$conjunction \rightarrow evidentiality$
(31) Oddrún er <u>týðiliga</u> tó klænkað.Oddrún has obviously though lost weight.	3%	10%	87%	evidentiality \rightarrow conjunction
(32) Hanus var tó <u>tíbetur</u> sloppin óskaddur. Hanus was though fortunately escaped unharmed.	78%	11%	11%	conjunction \rightarrow evaluation
(33) Hanus var <u>tíbetur</u> tó sloppin óskaddur. Hanus was fortunately though escaped unharmed.	23%	30%	47%	evaluation \rightarrow conjunction

 Table 6 The relative order of two adverbs that convey conjunction, evaluation and evidentiality

Much like the Icelandic speakers, the Faroese speakers are far more approving of placing the conjunctive adverb before both the speech act adverb and the evidential adverb. This difference is in fact even more distinct in Faroese. Table 7 depicts a sentence pair with three central adverb phrases:

	32 spe	32 speakers			
	Yes	?	No	Order	
 (34) Maria hevur satt at siga <u>tíbetur</u> týðiliga lisið bókina. Maria has honestly fortunately obviously read the book. 	41%	31%	28%	speech act \rightarrow evaluation \rightarrow evidentiality	
 (35) Maria hevur satt at siga týðiliga tíbetur lisið bókina. Maria has honestly obviously fortunately read the book. 	3%	3%	94%	speech act \rightarrow evidentiality \rightarrow evaluation	

 Table 7 The relative order of three adverbs that convey a speech act, evaluation and evidentiality

We can also see a much more distinct difference in word order variations than in the Icelandic data. While example (34) is refuted by the majority of the Faroese speakers, it is anyway much better received than (35) which is deemed impossible by most.

Finally, *Table 8* shows examples of different relative orders of a logical subject on one hand and different kinds of central adverbs on the other:

Table 8 The relative order of a sentence adverb and a logical subject

	32 speakers			
	Yes	?	No	Order
(36) Tað hava satt at siga <u>nógv</u> lisið bókina. EXPL have honestly many read the book.	58%	29%	13%	speech act adverb → subject
(37) Tað hava <u>nógv</u> satt at siga lisið bókina. EXPL have many honestly read the book.	19%	23%	58%	subject \rightarrow speech act adverb
(38) Tað hava tíbetur <u>nógv</u> lisið bókina. EXPL have fortunately many read the book.	65%	16%	19%	evaluative adverb \rightarrow subject
(39) Tað hava <u>nógv</u> tíbetur lisið bókina. EXPL have many fortunately read the book.	19%	6%	75%	subject \rightarrow evaluative adverb
 (40) Tað hevur helst <u>onkur útlendingur</u> keypt húsið hjá Eivindi. EXPL has probably some foreigner bought Eivindur's house 	71%	16%	13%	evidential adverb → subject
 (41) Tað hevur <u>onkur útlendingur</u> helst keypt húsið hjá Eivindi. EXPL has some foreigner probably bought Eivindur's house 	10%	6%	84%	subject → evidential adverb

In these examples, the order Adverb-Subject is always better received, just like in the Icelandic questionnaire. The order Subject-Adverb is however usually very poorly received. In this regard, the rules on word order seem to be more rigid in Faroese than in Icelandic. We should however keep in mind that the sentence examples were randomly set up in this part of the Icelandic questionnaire (UA) while the Faroese questionnaire (and the UI questionnaire) had responding minimal pairs or three sentences of a kind that presented a direct comparison (see discussion about the use of questionnaires in syntax research in Thráinsson et al. 2013).

4. Conclusion

The data presented in this paper show that the relative orders of central sentence adverbs that follow Cinque's (1999) hierarchy are generally more positively received in the Scandinavian Insular languages then the orders who do not follow the hierarchy. Examples that present three central sentence adverbial phrases are generally rather poorly received in both languages (perhaps due to difficulty in interpretation) but the main pattern seems to follow Cinque's hierarchy nonetheless. Examples of double subject constructions with a logical subject preceding a sentence adverb are usually far worse received in Faroese than in Icelandic. This indicates that the restraints on word order are more rigid in Faroese than in Icelandic. The difference might however be explained to some extent with regard to the fact that this part of the Icelandic questionnaire had random sentence examples while the Faroese questionnaire had speakers comparing minimal pairs side by side and three sentences of a kind. Further research is thus needed.

References

- Alexiadou, Artemis. 1997. *Adverb Placement. A case study in antisymmetric syntax.* Benjamins, Amsterdam.
- Angantýsson, Ásgrímur. 2017. Setningaratviksorð í íslensku og færeysku. *Íslenskt mál* 39: 75–86.
- Angantýsson, Ásgrímur. 2018. Verb-second in embedded clauses in Faroese. *Studia Linguistica* 72 (1): 165–189.
- Cinque, Guglielmo. 1999. Adverbs and functional heads: A cross-linguistic perspective. Oxford University Press, Oxford.
- Cinque, Guglielmo. 2004. Issues in adverbial syntax. Lingua 114(6): 683-710.
- Delfitto, Denis. 2006. Adverb Classes and Adverb Placement. H. v. Riemsdijk og M. Everaert (eds.): *The Blackwell Companion to Syntax*, I. vol., pp. 83–120. Blackwell, Oxford.
- Ernst, Thomas. 2002. The Syntax of Adjuncts. Cambridge University Press, Cambridge.
- Ernst, Thomas. 2004. Principles of adverbial distribution in the lower clause. *Lingua* 114(6): 755–777.
- Ernst, Thomas. 2007. On the role of semantics in a theory of adverb syntax. *Lingua* 117(6): 1008–1033.

- Jackendoff, Ray. 1972. Semantic Interpretation in Generative Grammar. MIT Press, Cambridge, MA.
- Jónsson, Jóhannes Gísli. 1996. *Clausal Architecture and Case in Icelandic*. GLSA, University of Massachusetts, Amherst.
- Jónsson, Jóhannes Gísli. 2002. S-Adverbs in Icelandic and the Feature Theory of Adverbs. *Leeds Working Papers in Linguistics and Phonetics* 9: 73–89.
- Jóhannsdóttir, Kristín. 2005. Temporal Adverbs in Icelandic: Adverbs of Quantification vs. Frequency Adverbs. Working Papers in Scandinavian Syntax 76: 31–72.

Nielsen, Øystein. 2000. The Syntax of Circumstantial Adverbs. Novus: Oslo.

- Pittner, Karin, Daniela Elsner og Fabian Barteld (eds.). 2015. Adverbs. Functional and diachronic aspects. Amsterdam: John Benjamins.
- Sveinn Bergsveinsson. 1969. Die Stellung des Adverbs im Isländischen. Folia Linguistica 3:307–332.
- Svenonius, Peter. 2002. Subject Positions and the Placement of Adverbials. P. Svenonius (eds.): *Subjects, Expletives and the EPP*, bls. 202–242. Oxford University Press, Oxford.
- Thráinsson, Höskuldur. 2005. *Íslensk tunga III. Setningar. Handbók um setningafræði.* Almenna bókafélagið, Reykjavík.
- Thráinsson, Höskuldur, Ásgrímur Angantýsson, Þórhallur Eyþórsson, Einar Freyr Sigurðsson and Sigrún Steingrímsdóttir. 2013. 'Efnissöfnun og aðferðafræði'. Höskuldur Thráinsson, Ásgrímur Angantýsson, og Einar Freyr Sigurðsson eds.): *Tilbrigði í íslenskri setningagerð I. Yfirlit yfir aðferðir og helstu niðurstöður*, pp. 19–68. Reykjavík: Háskólaútgáfan.
- Thráinsson, Höskuldur. 2007. *The Syntax of Icelandic*. Cambridge University Press, Cambridge.
- Thráinsson, Höskuldur. 2010. Predictable and unpredictable sources of variable verb and andverb placement in Scandinavian. *Lingua* 120: 1062-1088.
- Travis, Lisa D. 1988. The Syntax of Adverbs. McGill Working Papers, Special Issue on Comparative Germanic Syntax, bls. 280–310. Department of Linguistics, McGill University, Montreal.

Ásgrímur Angantýsson University of Iceland Árnagarði v. Suðurgötu IS-101 Reykjavík, Iceland asgriman@hi.is

Working Papers in Scandinavian Syntax

These working papers have been sponsored by the Norwegian Research Council for Science and the Humanities (NAVF) (no. 1-27) and by the Swedish Research Council for the Humanities and the Social Sciencies (HSFR) (no. 28-42), as well as by Erik Philip-Sörensen's stiftelse (no. 42-43). Issues 80-92 were sponsored by the Centre for Languages and Literature, Lund University. Issues 93-99 were published by Ghent University and Lund University. As of issue 100, WPSS is published by Stockholm University and Lund University.

PUBLISHED BY JUNE 2019

- Tarald Taraldsen: Som (1983) 1.
- 2. Christer Platzack: Germanic word order and the COMP/INFL parameter (1983)
- Anders Holmberg: The finite sentence in Swedish and English (1983) 3.
- Kirsti Koch Christensen: The categorial status of Norwegian infinitival relatives (1983) 4.
- 5. Lars Hellan: Anaphora in Norwegian and theory of binding (1983)
- Elisabet Engdahl: Parasitic gaps, subject extractions, and the ECP (1983) 6.
- 7. Elisabet Engdahl: Subject gaps (1984)
- Eiríkur Rögnvaldsson: Icelandic word order and flað-insertion 8. Höskuldur Thráinsson: Some points on Icelandic word order (1984)
- 9. Tarald Taraldsen: Some phrase structure dependent differences between Swedish and Norwegian (1984)
- Jan Engh: On the development of the complex passive 10. Lars Hellan: A GB-type analysis of complex passives and related constructions (1984)
- Tor A. Åfarli: Norwegian verb particle constructions as causative constructions (1984)
- 11. 12. Martin Everaert: Icelandic long reflexivization and tense-connectedness (1984)
- Anders Holmberg: On raising in Icelandic and Swedish 13.
- Anders Holmberg: On certain clitic-like elements in Swedish (1984)
- 14. Toril Fiva: NP-internal chains in Norwegian (1984)
- Kirsti Koch Christensen: Subject clitics and A-bound traces (1984) 15.
- Annie Zaenen, Joan Maling, Höskuldur Thráinsson: Passive and oblique case 16. Joan Maling, Annie Zaenen: Preposition-stranding and oblique case (1984)
- 17. Nomi Erteschik-Shir: Der (1985)
- 18. Halldór Ármann Sigurðsson: Subordinate V/I in Icelandic. How to explain a root phenomenon (1985)
- Kirsti Koch Christensen: Complex passive and conditions on reanalysis (1985) 19.
- 20. Christer Platzack: The Scandinavian languages and the null subject parameter (1985)
- 21. Anders Holmberg: Icelandic word order and binary branching (1985)
- Tor A. Åfarli: Absence of V2 effects in a dialect of Norwegian (1985) 22.
- 23. Sten Vikner: Parameters of binder and of binding category in Danish (1985)
- Anne Vainikka: Icelandic case without primitive grammatical functions (1985) 24.
- 25. Halldór Ármann Sigurðsson: Moods and (long distance) reflexives in Icelandic (1986)
- Wim Kosmeijer: The status of the finite inflection in Icelandic and Swedish (1986) 26.
- 27. Robin Cooper: Verb second - predication or unification? (1986)
- 28. Joan Maling: Existential sentences in Swedish and Icelandic: Reference to Thematic Roles (1987)
- 29. Tor A. Åfarli: Lexical structure and Norwegian passive and ergative constructions (1987)
- Kjell-Åke Gunnarsson: Expressions of distance and raising (1987) 30.
- Squibs, Remarks and Replies (Klaus von Bremen, Christer Platzack) (1987) 31.
- 32. Cecilia Falk: Subjectless clauses in Swedish (1987)
- 33. Anders Holmberg: The Structure of NP in Swedish (1987)
- 34. Halldor Ármann Sigurðsson: From OV to VO: Evidence from Old Icelandic (1988)
- 35. Lars Hellan: Containment and Connectedness Anaphors (1988)
- 36. Tomas Riad: Reflexivity and Predication (1988)
- 37. Squibs, Remarks and Replies (Elly van Gelderen, Arild Hestvik, Tomas Riad) (1988)
- 38. Sten Vikner & Rex A. Sprouse: Have/Be-Selection as an A-Chain Membership Requirement. (1988)
- Sten Vikner: Modals in Danish and Event Expressions (1988) 39
- Elisabet Engdahl: Implicational Universals: Parametric Variation in GB and GPSG. (1988) 40.
- Kjell-Åke Gunnarsson: Expressions of Distance, Prepositions and Theory of Theta-Roles (1988) 41.

Beginning with no. 42, the papers were no longer published as separate issues. There are two issues each year, one in June and one in December.

42. [December 1988]

Lars Hellan: The Phrasal Nature of Double Object Clusters

Anders Holmberg & Christer Platzack: On the Role of Inflection in Scandinavian Syntax

Barbro Lundin & Christer Platzack: The Acquisition of Verb Inflection, Verb Second and Subordinate Clauses in Swedish

Lars Olof Delsing: The Scandinavian Noun Phrase

Gunnel Källgren & Ellen F. Prince: Swedish VP-Topicalization and Yiddish Verb-Topicalization

43. [June 1989]

Torbjørn Nordgård: On Barriers, Wh-movement and IP-Adjunction in English, Norwegian and Swedish Bonnie D.Schwartz & Sten Vikner: All Verb Second Clauses are CPs. Christer Platzack & Anders Holmberg: The Role of AGR and Finiteness.

44. [December 1989]

Special Issue on Comparative Germanic Syntax

Tor Åfarli: On Sentence Structure in Scandinavian Languages.

Jan Anward: Constraints on Passives in Swedish and English.

Kathrin Cooper & Elisabet Engdahl: Null Subjects in Zurich German.

Cecilia Falk: On the Existential Construction in the Germanic Languages.

Lars Hellan: A Two Level X-bar System.

Jarich Hoekstra & Lásló Marácz: On the Position of Inflection in West-Germanic.

Kjartan G. Ottósson: VP-Specifier Subjects and the CP/IP Distinction in Icelandic and Mainland Scandinavian.

Charlotte Reinholtz: V-2 in Mainland Scandinavian: Finite Verb Movement to Agr.

Wolfgang Sternefeld: Extractions from Verb-Second Clauses in German.

Sten Vikner: Object Shift and Double Objects in Danish.

Chris Wilder: Wh-Movement and Passivization in Infinitive Predicates

45. [June 1990]

Helge Lødrup: VP-topicalization and the Verb *gjøre* in Norwegian.

Christer Platzack: A Grammar Without Functional Categories: A Syntactic Study of Early Swedish Child Language

Halldór Sigurðsson: Icelandic Case-marked PRO and the Licensing of Lexical A-positions.

46. [December 1990]

Halldór Sigurðsson: Feature Government and Government Chains Lena Ekberg: Theta Role Tiers and the Locative PP in Existential Constructions Sjur Nørstebø Moshagen & Trond Trosterud: Non-Clause-Bounded Reflexives in mainland Scandinavian Cecilia Falk: On Double Object Constructions

47. [June 1991]

Norbertt Hornstein: Expletives: a comparative study of English and Icelandic Lars-Olof Delsing: Quantification in the Swedish Noun Phrase Helge Lødrup: The Norwegian Pseudopassive in Lexical Theory Gunlög Josefsson: Pseudocoordination – A VP + VP Coordination

48. [December 1991]

Jóhannes Gísli Jónsson: Stylistic Fronting in Icelandic Kirsti Koch Christensen: Complex Passives Reanalyzed Kjartan G. Ottósson: Icelandic Double Objects as Small Clauses

49. [June 1992]

Halldór Sigurðsson: The Case of Quirky Subjects Anders Holmberg: Properties of Non-heads in Compounds: A Case Study Gunlög Josefsson: Object Shift and Weak Pronominals in Swedish Peter Svenonius: The Extended Projection of N: Identifying the Head of the Noun Phrase

50. [December 1992]

Sabine Iatridou and Anthony Kroch: The Licensing of CP-recursion and its Relevance to the Germanic Verb Second Phenomenon.

Christer Platzack: Complementizer Agreement and Argument Clitics. Halldór Sigurðsson: Agreement as Visible F-government. Tor A. Åfarli: Seeds and Functional Projections.

51. [June 1993]

Molly Diesing & Eloise Jelinek: The Syntax and Semantics of Object Shift.

52. [December 1993]

Gunlög Josefsson: Scandinavian Pronouns and Object Shift Anders Holmberg: Two Subject Positions in IP in Mainland Scandinavian

53. [June 1994]

Hans-Martin Gärtner & Markus Steinbach: Economy, Verb Second, and the SVO - SOV Distinction. Kyle Johnson & Sten Vikner: The Position of the Verb in Scandinavian Infinitives: In V° or C° but not in I°. Christer Platzack: Null Subjects, Weak Agr and Syntactic Differences in Scandinavian.

54. [December 1994]

Jan-Wouter Zwart: The Minimalist Program and Germanic Syntax. A Reply to Gärtner and Steinbach Knut Tarald Taraldsen: Reflexives, pronouns and subject / verb agreement in Icelandic and Faroese Christer Platzack: The Initial Hypothesis of Syntax: A Minimalist Perspective on Language Acquisition and Attrition

55. [June 1995]

Sten Vikner: V°-to-I° Movement and Inflection for Person in All Tenses

Anders Holmberg & Görel Sandström: Scandinavian Possessive Constructions from a Northern Swedish Viewpoint

Höskuldur Thráinsson and Sten Vikner: Modals and Double Modals in the Scandinavian Languages Øystein Alexander Vangsnes: Referentiality and Argument Positions in Icelandic

56. [December 1995]

Gunlög Josefsson: The Notion of Word Class and the Internal Make-up of Words Lars Hellan and Christer Platzack: Pronouns in Scandinavian Languages: An Overview Joan Maling and Jóhannes Gísli Jónsson: On Nominative Objects in Icelandic and the Feature [+Human]

57. [June 1996]

Halldór Ármann Sigurðsson: Icelandic Finita Verb Agreement Peter Svenonius: The Optionality of Particle Shift Helge Lødrup: The Theory of Complex Predicates and the Norwegian Verb *få* 'get' Thorbjörg Hróarsdóttir: The decline of OV Word Order in the Icelandic VP

58. [December 1996]

Øystein Alexander Vangsnes: The role of gender in (Mainland) Scandinavian possessive constructions Anna-Lena Wiklund: Pseudocoordination is Subordination Eiríkur Rögnvaldsson: Word Order Variation in the VP in Old Icelandic Tor A. Åfarli: An Argument for a Minimalist Construal of Case Licensing

59. [June 1997]

Øystein Nilsen: Adverbs and A-shift Kristin M. Eide & Tor A. Åfarli: A Predication Operator: Evidence and Effects Christer Platzack: A Representational Account of Restrictive and Non-Restrictive Relatives: The Case of Swedish

60. (December 1997)

Sten Vikner: The Interpretation of Object Shift, Optimality Theory, and Minimalism Jóhanna Barðdal: Oblique Subjects in Old Scandinavian Elisabet Engdahl: Relative Clause Extractions in Context Anders Holmberg: Scandinavian Stylistic Fronting: Movement of Phonological Features in the Syntax

61. [June 1998]

Verner Egerland: On Verb-Second Violations in Swedish and the Hierarchical Ordering of Adverbs Gunlög Josefsson & Christer Platzack: Short Raising of V and N in Mainland Scandinavian Christer Platzack: A Visibility Condition for the C-domain Gunlög Josefsson: On the Licensing and Identification of (Optionally) Null Heads in Swedish

62. [December 1998]

Cedric Boeckx: Agreement Constraints in Icelandic and Elsewhere. Jens Haugan: Right Dislocated 'Subjects' in Old Norse.

63. [June 1999]

Jan Terje Faarlund: The notion of oblique subject and its status in the history of Icelandic Elisabet Engdahl: Versatile Parasitic Gaps Benjamin Lyngfelt: Optimal Control. An OT perspective on the interpretation of PRO in Swedish Gunlög Josefsson: Non-finite root clauses in Swedish child language

64. [December 1999]

Inger Rosengren: Rethinking the Adjunct Maria Mörnsjö: Theories on the Assignment of Focal Accent as Applied to Swedish Jóhanna Barðdal: The Dual Nature of Icelandic Psych-Verbs Christer Platzack: The Subject of Icelandic Psych-Verbs: a Minimalist Account

65 [June 2000]

Inger Rosengren: EPP and the Post-finite Expletive Anders Holmberg: Expletives and Agreement in Scandinavian Passives Halldór Ármann Sigurðsson: The Locus of Case and Agreement Jóhanna Barðdal and Valeria Molnár: Passive in Icelandic – Compared to Mainland Scandinavian

66 [December 2000]

Halldór Ármann Sigurðsson: To be an oblique subject: Russian vs. Icelandic Marit Julien : Optional *ha* in Swedish and Norwegian Hjalmar P. Petersen: IP or TP in Modern Faroese Christer Platzack & Gunlög Josefsson: Subject Omission and Tense in Early Swedish Child Language

67 [June 2001]

Thórhallur Eythórsson: The Syntax of Verbs in Early Runic Jóhanna Barðdal & Thórhallur Eythórsson: The Evolution of Oblique Subjects in Scandinavian Gunlög Josefsson: The True Nature of Holmberg's Generalization Revisited – Once Again Halldór Ármann Sigurðsson: Case: abstract vs. morphological

68 [December 2001]

 Hubert Haider: How to Stay Accusative in Insular Germanic
 Gunnar Hrafn Hrafnbjargarson: An Optimality Theory Analysis of Agreement in Icelandic DAT-NOM Constructions.
 Nomi Erteschik-Shir P-syntactic motivation for movement: imperfect alignment in Object Shift
 Zeljko Boskovic: PF Merger in Scandinavian: Stylistic Fronting and Object Shift

Susann Fischer & Artemis Alexiadou: On Stylistic Fronting: Germanic vs. Romance Lars-Olof Delsing: Stylistic Fronting, Evidence from Old Scandinavian

69 [June 2002]

Line Mikkelsen: Reanalyzing the definiteness effect: evidence from Danish Verner Egerland: On absolute constructions and the acquisition of tense Peter Svenonius: Strains of Negation in Norwegian Anders Holmberg & Thorbjörg Hróarsdóttir: Agreement and movement in Icelandic raising constructions

70 [December 2002]

Joan Maling: Icelandic Verbs with Dative Objects Jóhanna Barðdal: "Oblique Subjects" in Icelandic and German Halldór Ármann Sigurðsson: Agree and Agreement: Evidence from Germanic

71 [June 2003]

Arthur Stepanov: On the "Quirky" Difference Icelandic vs. German: A Note of Doubt. Janne Bondi Johannessen: Negative Polarity Verbs in Norwegian. Verner Egerland: Impersonal Pronouns in Scandinavian and Romance. Erik Magnusson: Subject Omission and Verb Initial Declaratives in Swedish. Thórhallur Eythórsson & Jóhanna Barðdal: Oblique Subjects: A Germanic Inheritance!

72 [December 2003]

Ken Ramshøj Christensen: On the Synchronic and Diachronic Status of the Negative Adverbial *ikke/not*. Luis López: Complex Dependencies: the Person-Number restriction in Icelandic. Katarina Lundin-Åkesson: Constructions with *låta* LET, reflexives and passive -s –

a comment on some differences, similarities and related phenomena.

Thorbjörg Hróarsdóttir: Economy: On simplicity, default values and markedness in language acquisition and change.

Gunnar Hrafn Hrafnbjargarson: On Stylistic Fronting Once More Thórhallur Eythórsson & Jóhannes Gísli Jónsson: The Case of Subject in Faroese

73 [June 2004]

Øystein Alexander Vangsnes: On wh-questions and V2 across Norwegian dialects.

A survey and some speculations.

David Håkansson: Partial *wh*-movement in the history of Scandinavian Christer Platzack: Agreement and the Person Phrase Hypothesis

74 [December 2004]

Halldór Ármann Sigurðsson: Agree in Syntax, Agreement in Signs

Ute Bohnacker: Is V2 really that hard to acquire for second language learners?

On current universalist L2 claims and their empirical underpinnings

Johan Brandtler: Subject Omission and Discourse Anchorage in Early Swedish Child Language

75 [June 2005]

Johanna Barðdal & Thórhallur Eythórsson: Case and Control Constructions in

German, Faroese and Icelandic: Or How to Evaluate Marginally-Acceptable Data? Fredrik Heinat: Reflexives in a phase based syntax

Gunlög Josefsson: How could Merge be free and word formation restricted:

The case of compounding in Romance and Germanic

Christer Platzack: Uninterpretable features and EPP: a minimalist account of language build up and breakdown

76 [December 2005]

Björn Rothstein: Perfect parasitism in inferential contexts. On the inferential present perfect in Swedish. Kristín M. Jóhannsdóttir: Temporal adverbs in Icelandic: Adverbs of quantification vs. frequency adverbs. Katarina Lundin Åkesson: The multifunctional *ba* – A finiteness marker in the guise of an adverbial. Halldór Ármann Sigurðsson: Accusative and the Nom/Acc alternation in Germanic. Fredrik Heinat: A note on 'long object shift'.

77 June [2006]

Marit Julien: On argument displacement in English and Scandinavian

Christer Platzack: Case as Agree Marker

Halldór Ármann Sigurðsson: PF is more 'syntactic' than often assumed

Jackie Nordström: Selection through Uninterpretable Features. Evidence from Insular Scandinavian

Camilla Thurén: The syntax of Swedish present participles. The lexical category problem.

Johan Brandtler: On Aristotle and Baldness - Topic, Reference, Presupposition of Existence, and Negation

78 December [2006]

Þorbjörg Hróarsdóttir, Gunnar Hrafn Hrafnbjargarson, Anna-Lena Wiklund and Kristine Bentzen: The Tromsø guide to Scandinavian verb movement.

Terje Lohndal: The phrase structure of the copula.

Ute Bohnacker: Placing verbs and particles in non-native German and Swedish.

Björn Rothstein: Why the present perfect differs cross linguistically. Some new insights.

Henrik Rosenkvist: Null subjects in Övdalian.

Piotr Garbacz: Verb movement and negation in Övdalian.

79 [June 2007]

Geoffrey Poole: Defending the "Subject Gap" Requirement: Stylistic Fronting in Germanic and Romance Jan Terje Faarlund: From clitic to affix: the Norwegian definite article

Terje Lohndal: *That*-t in Scandinavian and elsewhere: Variation in the position of C

Tor A. Åfarli: Features and Agreement. Expletive det 'it' and der 'there' in Norwegian dialects

Kristine Bentzen, Gunnar Hrafn Hrafnbjargarson, Þorbjörg Hróarsdóttir and Anna-Lena Wiklund: The Tromsø guide to the Force behind V2

Kristine Bentzen, Gunnar Hrafn Hrafnbjargarson, Þorbjörg Hróarsdóttir and Anna-Lena Wiklund: Extracting from V2

80 December [2007]

Željko Bošković: Don't feed your movements: Object shift in Icelandic

Werner Abraham & Elisabeth Leiss: On the interfaces between (double) definiteness,

aspect, and word order in Old and Modern Scandinavian

Þorbjörg Hróarsdóttir, Anna-Lena Wiklund, Kristine Bentzen & Gunnar Hrafn Hrafnbjargarson:

The afterglow of verb movement

Henrik Rosenkvist: Subject Doubling in Oevdalian

Marit Julien: Embedded V2 in Norwegian and Swedish

Britta Jensen: In favour of a truncated imperative clause structure: evidence from adverbs Mai Tungset: Benefactives across Scandinavian

81 [June 2008]

Halldór Ármann Sigurðsson & Joan Maling: Argument drop and the Empty Left Edge Condition (ELEC) Gunlög Josefsson: Pancakes and peas – on apparent disagreement and (null) light verbs in Swedish Fredrik Heinat: Long object shift and agreement Johan Brandtler: On the Structure of Swedish Subordinate Clauses

82 December [2008]

Elly van Gelderen & Terje Lohndal: The position of adjectives and double definiteness Terje Lohndal, Mari Nygård & Tor A. Åfarli: The structure of copular clauses in Norwegian Þorbjörg Hróarsdóttir: Verb particles in OV/VO word order in Older Icelandic Johan Brandtler: Why we should ever bother about *wh*-questions. On the NPI-licensing

properties of wh- questions in Swedish

Gunnar Hrafn Hrafnbjargarson: Liberalizing modals and floating clause boundaries

Tavs Bjerre, Eva Engels, Henrik Jørgensen & Sten Vikner: Points of convergence between functional and formal approaches to syntactic analysis.

83 [June 2009]

Ulla Stroh-Wollin: On the development of definiteness markers in Scandinavian. Anna-Lena Wiklund: In search of the force of dependent V2: A note on Swedish. Þorbjörg Hróarsdóttir: Restructuring and OV order.

Eva Engels: Microvariation in object positions: Negative Shift in Scandinavian. Þorbjörg Hróarsdottir: Notes on language change and grammar change. Dennis Ott: Stylistic fronting as remnant movement.

84 [December 2009]

Maia Andreasson: Pronominal object shift – not just a matter of shifting or not Gunnar Hrafn Hrafnbjargarson & Anna-Lena Wiklund: General embedded V2: Icelandic A, B, C, etc. Gunlög Josefsson: "Disagreeing" pronominal reference and gender in Swedish David Petersson: Embedded V2 does not exist in Swedish Henrik Rosenkvist: Referential null-subjects in Germanic languages – an overview Anna-Lena Wiklund: The syntax of Surprise: unexpected event readings in complex predication Marit Julien: The force of the argument Anna-Lena Wiklund: May the force be with you: A reply from the 5th floor

85 [June 2010]

Mayumi Hosono: Scandinavian Object Shift as the cause of downstep Jackie Nordström: The Swedish *så*-construction, a new point of departure Anton Karl Ingason: Productivity of non-default case

86 [December 2010]

Gunlög Josefsson; Object Shift and optionality. An intricate interplay between

syntax, prosody and information structure

Mayumi Hosono: On Icelandic Object Shift

Mayumi Hosono: Why Object Shift does not exist in Övdalian.

Mayumi Hosono: On Unshifted Weak Object Pronouns in the Scandinavian Languages.

Eva Engels: Local licensing in Faroese expletive constructions.

Irene Franco: Issues in the syntax of Scandinavian embedded clauses.

David Petersson & Gunlög Josefsson: ELLERHUR and other Yes/No-question operator candidates in Swedish.

Mikko Kupula: Causers as derived Subject - An unaccusative view from Finnish

87 [June 2011]

Jim Wood: Icelandic *let*-causatives and Case.

Eva Klingvall: On past participles and their external arguments.

Ulla Stroh-Wollin: Embedded declaratives, assertion and swear words.

Verner Egerland: Fronting, Background, Focus: A comparative study of Sardinian and Icelandic.

Caroline Heycock, Antonella Sorace, Zakaris Svabo Hansen, Sten Vikner & Frances Wilson:

Residual V-to-I in Faroese and its lack in Danish: detecting the final stages of a syntactic change.

88 [December 2011]

Henrik Rosenkvist; Verb Raising and Referential Null Subjects in Övdalian Kari Kinn: Overt non-referential subjects and subject-verb agreement in Middle Norwegian Mayumi Hosono: Verb Movement as Tense Operator Movement Jim Wood & Einar Freyr Sigurðsson: Icelandic Verbal Agreement and Pronoun Antecedent Relations Eva Klingvall: On non-copula *Tough* Constructions in Swedish David Petersson: Swedish exclamatives are subordinate

89 [June 2012]

Eva Engels: Wh-phrases and NEG-phrases in clauses and nominals. Fredrik Heinat: Adjective and clausal complementation. Mayumi Hosono: Information structure, syntax and information properties of multiple *Wh*-questions.

90 [December 2012]

Ermenegildo Bidese, Andrea Padovan, AlessandraTomaselli: A binary system of

complementizers in Cimbrian relative clauses Camilla Thurén: The syntax of Swedish copular clauses Eva Klingvall: Topics in pseudo passives Fredrik Heinat: Finiteness in Swedish. Gunlög Josefsson: "Disagreeing" doubling *det*

91 [December 2013]

Roland Hinterhölzl: Economy conditions and coreference: From minimal pronouns to referential acts Dorian Roehrs: Possessives as Extended Projections Björn Lundquist: On inter-individual variation and mid-distance binding in Swedish Verner Egerland: The Apropos-Topic, the Concerning-Topic and the syntax-pragmatics interface

92 [June 2014]

Elisabet Engdahl & Filippa Lindahl: Preposed object pronouns in Mainland Scandinavian Katarina Lundin: An unexpected gap with unexpected restrictions Dennis Ott: Controlling for movement: Reply to Wood (2012) Halldór Ármann Sigurðsson: About pronouns

93 [December 2014]

Filippa Lindahl: Relative Clauses are not always strong islands Gunlög Josefsson: Pseudo-coordination with gå 'go' and the "surprise effect" Jóhanna Barðdal, Thórhallur Eythórsson & Tonya Kim Dewey: Alternating Predicates in Icelandic and German Mayumi Hosono: Scandinavian Verb Particle Constructions and the Intonational Principles

94 [June 2015]

Marit Julien: Microvariation in Norwegian long distance binding Fredrik Heinat & Anna-Lena Wiklund: Scandinavian Relative Clause Extractions Mayumi Hosono: On Verb Movement in the *Labeling Algorithm*-Based Derivation

95 [December 2015]

Jan Terje Faarlund: The Norwegian infinitive marker Ulla Stroh-Wollin: Understanding the gradual development of definiteness marking: the case of Swedish Martje Wijers: Forgotten factors in the development of dependent clauses in Swedish as a second language

96 [June 2016]

Jim Wood: How roots do and don't constrain the interpretation of Voice Anton Karl Ingason, Einar Freyr Sigurðsson & Jim Wood: Displacement and subject blocking in verbal idioms Jim Wood, Matthew Barros & Einar Freyr Sigurðsson: Clausal ellipsis and case (mis)matching in Icelandic Thórhallur Eythórsson & Sigríður Sæunn Sigurðardóttir: A brief history of Icelandic weather verbs Ásgrímur Angantýsson & Dianne Jonas: On the syntax of adverbial clauses in Icelandic

97 [December 2016]

Hans-Martin Gärtner: A note on the Rich Agreement Hypothesis and varieties of "Embedded V2" Verner Egerland & Dianne Jonas: *Enough already*! On directive modal particles in English and Swedish Mayumi Hosono: Exceptional movement from/into the Criterial Position

Anton Karl Ingason, Iris Edda Nowenstein & Einar Freyr Sigurðsson: The Voice-adjunction theory of 'by'phrases and the Icelandic impersonal passive

Jóhannis Gísli Jónsson: Testing agreement with nominative objects

Special Issue on Icelandic

98 [June 2017]

Christer Platzack & Inger Rosengren: What makes the imperative clause type autonomous? A comparative study in a modular perspective.

Ásgrímur Angantýsson: Subordinate V2 and verbal morphology in Övdalian

Tam Blaxter & David Willis: Pragmatic differentiation of negative markers in the early stages of Jespersen's cycle in North Germanic

Ingun Hreinberg Ingriðadóttir: Weight effects and heavy NP shift in Icelandic and Faroese

99 [December 2018]

Dennis Wegner: The exceptional status of the Swedish supine. On the parametric variation of past participial (non-)identity.

Heimir van der Feest Viðarsson: Grimm's "floating" datives. Applicatives and NP/DP configurationality in Icelandic from a diachronic perspective.

Ásgrímur Angantýsson: The distribution of embedded V2 and V3 in modern Icelandic

100 [June 2018]

Cecilia Falk: From impersonal to passive verb.

Eric Lander: Revisiting the etymology of the Norse negative clitic -a/-at.

Mayumi Hosono: Constraints on movement.

Joachim Kokkelmans: Elvis Presley, God and Jane: the Germanic proprial article in a contrastive perspective.

101 [December 2018]

Elisabet Engdahl, Joan Maling, Halldór Ármann Sigurðsson & Annie Zaenen: Presentational sentences in Icelandic and Swedish – Roles and positions.

Anders Holmberg: The syntax of the V3 particle *så* in the Swedish left periphery.

Halldór Ármann Sigurðsson: Icelandic declarative V1: a brief overview

102 [June 2019]

Sígriður Sæunn Sigurðardóttir: Syntax and Discourse: Case(s) of V3 orders in Icelandic with temporal adjuncts. Christopher D. Sapp: Relative *sá* and the dating of Eddic and skaldic poetry.

Issues 1–43, 45, 66, 67 are out of stock. It is still possible to get copies of 44, 46–65, 68–80 by sending an order to the editor. Beginning with issue 81 (June 2008), the articles published in WPSS are available online: http://projekt.ht.lu.se/grimm/working-papers-in-scandinavian-syntax/